# SEXUAL AND REPRODUCTIVE HEALTH AND RIGHTS IN EUROPE PROGRESS AND CHALLENGES



Follow-up report

to the 2017 Issue Paper





# SEXUAL AND REPRODUCTIVE HEALTH AND RIGHTS IN EUROPE PROGRESS AND CHALLENGES

Follow-up report to the 2017 Issue Paper by the Council of Europe Commissioner for Human Rights Issue papers and their followup reports are published by the Commissioner for Human Rights to contribute to debate and reflection on important current human rights issues. Many of them also include recommendations by the Commissioner for addressing the concerns identified.

The opinions expressed in this followup report are the responsibility of the authors and do not necessarily reflect the official policy of the Council of Europe.

All requests concerning the reproduction or translation of all or part of this document should be addressed to the Directorate of Communication (publishing@coe.int).
All other correspondence concerning this document should be addressed to the Office of the Commissioner for Human Rights (commissioner@coe.int).

Issue Papers are available on the Commissioner's website: www.commissioner.coe.int

Cover photo: @Shutterstock

©Council of Europe, February 2024

#### Acknowledgements:

This follow-up report was prepared by Adriana Lamačková and Keina Yoshida, with assistance from Camille Gervais, Leah Hoctor, Laura Mirahver and Katrine Thomasen, from the Europe Program at the Center for Reproductive Rights.

## **Contents**

| Introd | uction   | 5  |
|--------|--|----|
| Chapte | er 1 - Comprehensive sexuality education   | 9  |
| 1.1    | Developments in international human rights standards and technical guidelines        | 9  |
| 1.2    | Deficits in establishing and delivering CSE  | 10 |
| 1.3    | Misconceptions and opposition to CSE   | 11 |
| Chapte | er 2 - Access to contraceptive services and information                              | 13 |
| Chapte | er 3 - Access to abortion care and information                                       | 15 |
| 3.1    | Developments in international human rights standards and public health guidelines    | 15 |
| 3.2    | Highly restrictive abortion laws   | 18 |
| 3.3    | Criminalisation of abortion  | 18 |
| 3.4    | Access barriers  | 19 |
| 3.5    | Access to evidence-based information   | 20 |
| 3.6    | Retrogression  | 20 |
| Chapte | er 4 - Obstetric and gynaecological care   | 23 |
| 4.1    | Developments in international human rights standards and public health guidelines    | 23 |
| 4.2    | Disrespectful care and mistreatment  | 25 |
| 4.3    | Informed consent and decision making   | 26 |
| -      | er 5 - Access to sexual and reproductive health and rights at discrimination         | 27 |
| 5.1    | Developments in international human rights standards                                 | 27 |
| 5.2    | Challenges and deficits  | 28 |
| _      | er 6 - Human rights defenders working to advance sexual and uctive health and rights | 31 |
| 6.1    | Risks to personal safety and fear of prosecution and judicial harassment             | 32 |
| 6.2    | Unfavourable and hostile working environments  | 32 |
| The Co | mmissioner's recommendations   | 35 |
| Endno  | tes  | 41 |

### Introduction

Universal access to quality sexual and reproductive health care and the ability to exercise sexual and reproductive autonomy and decision making are integral components of human rights and gender equality. Council of Europe member states have international human rights obligations to respect, protect and fulfil these rights.

In 2017, the Council of Europe Commissioner for Human Rights' predecessor published an Issue paper on *Women's sexual and reproductive health and rights in Europe* ("the 2017 Issue Paper"). The 2017 Issue Paper set out key international human rights standards relating to women's sexual and reproductive health and rights and described a series of key concerns, challenges, and deficits pertaining to women's and girls' sexual and reproductive health and rights in Council of Europe member states. The Issue Paper also outlined recommendations for action by member states to ensure the full enjoyment of sexual and reproductive health and rights.

Since 2017, member states have taken important steps towards the implementation of these recommendations. Many countries have adopted legal and policy reforms, upgraded their programmatic, investment and health system responses, enhanced the levels of technical guidance and training provided to health care workers, and improved oversight, accountability, and monitoring mechanisms. These developments have increased access to contraception and abortion care, improved the quality of obstetric care and advanced the provision of comprehensive sexuality education in schools, amongst other things.

However, despite many notable achievements, pervasive challenges and deficits in sexual and reproductive health and rights continue across Europe and there is an ongoing need for robust member state action to ensure the full enjoyment of sexual and reproductive health and rights for everyone. As the Commissioner has outlined in numerous reports, statements, and comments published since 2017, the following problems still persist: human rights violations continue to occur in the sphere of sexual and reproductive health; retrogression has taken place in some member states; multiple and intersectional discrimination continues to impede the full realisation of sexual and reproductive health and rights of marginalised groups; and

human rights defenders working to advance sexual and reproductive health and rights still face considerable challenges in parts of the region.

In light of the new landscape in the Council of Europe region following the COVID-19 pandemic, the full-scale invasion of Ukraine by the Russian Federation in 2022 and the related cost-of-living crisis, this new report, which includes the Commissioner's follow-up recommendations, reiterates the importance of adopting an inclusive approach in the design and implementation of sexual and reproductive health policies. It stresses the need to pay special attention to women and girls in all their diversity\* as well as non-binary people and other persons with relevant needs, such as transgender (trans) men, who may be at risk of, or exposed to, intersecting forms of discrimination, and who might face increased vulnerabilities.

This report is a follow-up to, and complements, the 2017 Issue Paper and is intended to assist member states with their ongoing efforts to improve the enjoyment of sexual and reproductive health and rights.

The report highlights some of the concerns in certain areas of sexual and reproductive health and rights and provides a concise overview of important recent developments in relevant international human rights standards and international public health and technical guidelines. It draws on the Commissioner's country work, expert meetings with civil society organisations and dialogue with member states.

Section 1 focuses on the provision of comprehensive sexuality education (CSE) in Council of Europe member states. It highlights some of the prevailing concerns and deficits in relation to the provision of CSE and points to recent developments in international human rights standards and technical guidelines.

Section 2 addresses access to contraception services and information, and outlines some of the progress and ongoing challenges and recent developments in international human rights standards.

Section 3 considers the state of access to abortion care and information in the region and discusses developments in international human rights law and public health standards on abortion. It identifies the important progress made by many Council of Europe member states since the 2017 Issue Paper whilst also highlighting the impact of remaining barriers and restrictions that continue to undermine access to quality abortion care.

Section 4 addresses mistreatment and abuse in reproductive health care.

<sup>\*</sup> This publication, when using "women and girls", refers to all women and girls regardless of their age, disability, gender identity, gender expression, racial or ethnic origin, religion or belief, sex characteristics, sexual orientation, or other characteristics or status.

It outlines important developments in international human rights law and public health standards and provides a brief overview of the main issues and concerns that continue to impede the delivery of respectful and dignified obstetric and gynaecological care in many member states.

Section 5 addresses the need for urgent action to ensure that women and girls can fully enjoy their sexual and reproductive health and rights free from all forms of discrimination. It illustrates the way in which intersectional discrimination continues to affect the sexual and reproductive health and rights of many marginalised groups across Europe, with a particular focus on women and girls with disabilities, Roma women and girls, older women, refugees, asylum seeking and migrant women, LGBTI people and sex workers.

Section 6 highlights the vital role of human rights defenders, civil society organisations and health care providers working to advance sexual and reproductive health and rights and summarises some of the challenges and risks they are facing due to their work.

This follow-up report is followed by the Commissioner's recommendations to all Council of Europe member states in the areas of sexual and reproductive health and rights addressed in the report.

## Chapter 1

# **Comprehensive sexuality education**

Comprehensive sexuality education (CSE) plays a vital role in promoting the health and wellbeing of children and young people and preparing them for safe, fulfilling lives.<sup>2</sup> Ensuring that children and young people have access to age-appropriate CSE is a central component of states' obligations to guarantee sexual and reproductive health and rights. Data and evidence clearly outline that good quality, curriculum-based CSE enables children and young people to develop age-appropriate knowledge, attitudes, and skills that contribute to safe, healthy, and positive relationships.<sup>3</sup>

In recent years, some member states have made important progress towards ensuring the provision of age-appropriate, evidence-based CSE in school settings. However, in many parts of the region, deficits and challenges in CSE programming and implementation persist, and harmful misconceptions about CSE remain at play, depriving many children and young people of access to this important form of education.

## 1.1 Developments in international human rights standards and technical guidelines

International human rights law requires member states to guarantee access to age-appropriate CSE for all children and young people and to prevent retrogressive initiatives that would restrict or remove access to CSE.<sup>4</sup>

In recent years, human rights mechanisms have continued to emphasise the importance of comprehensive, evidence-based, scientifically accurate, age-appropriate and quality sexuality education.<sup>5</sup> They have made it clear that international human rights law requires member states to ensure that comprehensive sexuality education is non-discriminatory and inclusive.<sup>6</sup> They have outlined that CSE must promote human rights, gender equality, respect, autonomy, consent and diversity by supporting the development of skill sets that will enable young people to: exercise informed decision

making about sexuality; protect themselves from sexual and gender-based violence, sexually transmitted infections and early pregnancy; and build healthy and respectful relationships. They have underscored that member states must address information accessibility in digital environments, including guidance on how to identify trusted sources of information, and they must also ensure that CSE teachers are trained on digital safeguards. In addition, human rights mechanisms have underlined that states should integrate age-appropriate, gender-sensitive and accessible CSE into mandatory school curricula at all levels of education and provide systematic training to teachers.

New International technical guidance on comprehensive sexuality education has also been published in recent years, which outlines a new definition of CSE and contains key considerations and recommendations for member states regarding CSE.<sup>10</sup> The guidance has defined CSE as "a curriculumbased process of teaching and learning about the cognitive, emotional, physical, and social aspects of sexuality. It aims to equip children and young people with knowledge, skills, attitudes, and values that will empower them to: realise their health, wellbeing and dignity; develop respectful social and sexual relationships; consider how their choices affect their own wellbeing and that of others; and understand and ensure the protection of their rights throughout their lives."<sup>11</sup>

#### 1.2 Deficits in establishing and delivering CSE

Although there has been some important progress, shortcomings in compliance with international standards and guidelines on CSE continue across Europe. Concerns persist regarding limited access to CSE in schools in many member states. Data indicates that several member states still have not integrated CSE into mandatory school curricula.<sup>12</sup>

Even where sexuality education is mandatory and supportive legislative or policy frameworks exist, there are still notable variations in the quality of curricula, teacher training, and delivery methods. School curricula on sexuality education often falls short of international standards, with a strong focus on the biological aspects of sex and reproduction, and far less content on gender equality, sexuality, consent, human rights and digital media. Gaps also persist in addressing social norms and gender stereotypes, including those around toxic masculinity, and men and boys must be included in efforts towards gender equality and eliminating all forms of discrimination, sexism, misogyny, and violence. Children and young people are rarely involved in developing and evaluating CSE curricula and delivery.

Support for teachers on the delivery of sexuality education continues to be insufficient in most countries<sup>17</sup> and teachers continue to lack access to adequate training materials and tools and report a lack of confidence when it comes to addressing diverse topics, including gender norms, or in the use of interactive teaching methods.<sup>18</sup>

Inadequate resource allocation to CSE programming and monitoring and evaluation also remains a serious concern in most countries, and CSE implementation is often dependent on local authorities or individual schools.<sup>19</sup>

#### 1.3 Misconceptions and opposition to CSE

In some member states, misconceptions and disinformation about CSE persist and public campaigns have sometimes sought to disseminate harmful and inaccurate information about CSE, wrongfully describing it as "sexual abuse," or "paedophilia," or so-called "propaganda in favour of homosexuality." These campaigns also often claim that CSE is intended to spread so-called "gender ideology," and that mandatory CSE in school curricula violates the rights of parents to educate their children in accordance with their values and beliefs.<sup>20</sup>

In a small number of countries, policy makers have pursued efforts to roll back CSE programming or block the introduction of inclusive non-discriminatory CSE in schools. For example, in 2021, legislation was adopted in Hungary prohibiting or limiting children's access to any LGBTI content in the public sphere, including in schools and the media.<sup>21</sup>

## Chapter 2

## Access to contraceptive services and information

In recent years, several Council of Europe member states have made progress in removing barriers that impede access to affordable contraceptive services and information. Yet, for many across Europe, access to a choice of affordable modern contraceptive methods remains a challenge. The COVID-19 pandemic and associated measures aggravated pre-existing access barriers and at times created new ones, such as service disruptions, stock outages, increased costs, and limited access to medical consultations for prescription refills.<sup>22</sup>

Although some member states such as Belgium, France, Ireland, and Luxembourg have expanded entitlements to free contraception in recent years,<sup>23</sup> in many member states financial barriers and the failure to cover contraception under public health insurance or reimbursement schemes continue to undermine access for many women and girls.

Other ongoing barriers include the restricted availability of certain contraceptive options, prescription requirements, and the lack of access to evidence-based accessible information on a wide-range of contraceptive options. In many member states, myths and misconceptions about modern contraception persist due notably to insufficient training for health professionals and the lack of evidence-based public awareness campaigns.<sup>24</sup>

Women and girls from marginalised groups, including women and girls living in poverty, rural women and girls, women and girls with disabilities, ethnic minorities, refugees, asylum seekers and migrants, particularly those in an irregular situation, face specific challenges when it comes to gaining access to contraception. Laws and policies still prevent many adolescents in the region from gaining access to prescriptions for contraceptives since some member states still require parental consent for adolescents under the ages of 16 or 18 years. Limited engagement of men and boys as both users and supporters of contraception and family planning in a way that

promotes gender equality and supports women's sexual and reproductive decision making and autonomy also continues to be of concern across Europe.<sup>25</sup>

Although most member states have now legalised the sale of emergency contraception over the counter in pharmacies, without prescription, some still require a prescription for emergency contraception causing significant delays in access to time-sensitive medication.<sup>26</sup>

International human rights law requires member states to ensure access to evidence-based comprehensive contraceptive services and information, on the basis of non-discrimination and equality.<sup>27</sup> In recent years, human rights mechanisms have expanded on the content of these obligations, calling on member states to make the full range of modern contraception, including emergency contraception, readily available and accessible to everyone, including by covering them under national health insurance schemes or providing them free of charge.<sup>28</sup> They have underlined the importance of effective measures to disseminate comprehensive, evidence-based information on all modern contraceptive options in accessible formats, including through comprehensive sexuality education,<sup>29</sup> and have stressed that member states should "conduct awareness-raising programmes to promote the understanding that contraception is a responsibility that women and men share."30 They have also addressed the need to ensure that women and girls with disabilities are supported, where needed, in decision making concerning their use of contraception and that this must always be based on their free and informed consent, respecting their dignity and autonomy.31

### Chapter 3

# Access to abortion care and information

Following decades of legal and policy reforms intended to expand access to abortion care and improve service provision, abortion is now legal in almost all Council of Europe member states.

Since the publication of the 2017 Issue Paper, Cyprus, Finland, Iceland, Ireland, Northern Ireland and San Marino have legalised abortion on request,<sup>32</sup> meaning that doctors or other professionals are not required to attest to, or certify, the existence of a particular reason or justification for the abortion. Over the same period, many places in Europe, including Belgium, France, Lithuania, Moldova, the Netherlands, North Macedonia, Spain, as well as England, Scotland and Wales have undertaken reforms designed to remove some of the harmful procedural or regulatory barriers that have obstructed access to abortion care in practice, such as mandatory waiting periods, third-party authorisation requirements and restrictions on medication abortion.<sup>33</sup> Similar law reform processes that have the potential to remove other harmful barriers and restrictions are currently under way in numerous member states, including Belgium, Denmark, France, Germany, Ireland, the Netherlands, Norway, Poland, and the United Kingdom.<sup>34</sup>

However, there is sizeable room for improvement. Laws and policies on abortion continue to fall short of international human rights standards and public health guidance. Meanwhile, rollbacks on entitlements to abortion have occurred in a small number of countries and retrogressive threats continue to pose a risk to member states' commitments under international human rights law.

## 3.1 Developments in international human rights standards and public health guidelines

Guaranteeing access to quality abortion care is a critical component of states' obligations to respect and ensure the human rights of women and

other persons in need of such care, and human rights mechanisms have long recognised that the criminalisation of abortion and restrictive laws, policies and practices undermine access to safe abortion care and impose burdens on those seeking care, all of which can jeopardise their health and lives.<sup>35</sup>

In recent years, human rights mechanisms have continued to articulate the content of member states' obligations to decriminalise abortion, repeal restrictive abortion laws, and guarantee access to abortion care in practice. They have urged states to guarantee "adequate access to safe abortion and post-abortion services to ensure full realisation of rights of women, their equality and their economic and bodily autonomy to make free choices about their reproductive rights" and made it clear that any regulation of abortion must not subject women and girls who need abortion care to physical or mental pain or suffering, discriminate against them, arbitrarily interfere with their privacy or force them to resort to unsafe abortion.<sup>37</sup>

To this end they have specified that member states must:

- ✓ Legalise abortion, reform highly restrictive abortion laws that limit access to specific situations and decriminalise abortion in all circumstances, and repeal provisions that criminalise abortion and related assistance to which a pregnant person has given free and informed consent.<sup>38</sup>
- ✓ Remove access barriers such as mandatory waiting periods, mandatory counselling, and third-party authorisation requirements and ensure that abortion services are confidential and non-judgmental.<sup>39</sup>
- ✓ Guarantee universal and equal access to abortion care for everyone across their jurisdiction.<sup>40</sup> This should include covering abortion care, including abortion on request, under public health insurance schemes or providing it free of charge,<sup>41</sup> and ensuring that "sufficient numbers of adequately trained medical professionals are available to perform abortions and reduce regional disparities in this regard."<sup>42</sup>
- ✓ Guarantee the availability of medication abortion and ensure it is provided based on non-discrimination and equality.<sup>43</sup>
- Refrain from imposing new barriers in access to abortion care and ensure there is no retrogression in entitlements to care.44

Human rights mechanisms have addressed refusals of abortion care by medical professionals.<sup>45</sup> For example, in two 2020 decisions, the European Court of Human Rights considered complaints by two midwives against Sweden alleging that their right to freedom of thought, conscience and religion under Article 9 of the European Convention on Human Rights had

been violated when they were denied employment within the Swedish health system following refusals to assist in abortion care due to their personal, religious faith. The Court dismissed their claims as manifestly ill-founded, outlining that the denial of employment, "pursued the legitimate aim of protecting the health of women seeking an abortion," and was necessary in a democratic society and proportionate because, "Sweden provides nationwide abortion services and therefore has a positive obligation to organise its health system in a way as to ensure that the effective exercise of freedom of conscience of health professionals in the professional context does not prevent the provision of such services." 46

Similarly, human rights mechanisms have recognised that those seeking or providing abortion care may face stigmatisation and harassment due to state failures to address the actions of individuals and groups who oppose abortion care. They have called on states to "[s]trengthen efforts to prevent the stigmatisation and trauma of women and girls who seek abortion, including through the provision of "safe access zones" in all concerned health services in a timely manner."<sup>47</sup> They have also stressed the importance of measures to "enable health-care providers to undertake their work without undue interference, intimidation or restrictions."<sup>48</sup>

Public health guidelines on abortion care have also evolved in recent years. Drawing on international human rights standards and public health evidence, the World Health Organization (WHO) issued its Abortion Care Guideline in 2022, updating and replacing the recommendations in previous WHO guidelines on abortion care.<sup>49</sup> The Guideline explains that an enabling environment – which includes respect for human rights including a supportive framework of law and policy, the availability and accessibility of information, and a supportive, universally accessible, affordable and well-functioning health system – is the foundation of quality comprehensive abortion care.<sup>50</sup> The Guideline points out that legal, policy and regulatory restrictions, including limiting access to abortion by grounds and gestational limits, as well as other barriers, can make it difficult or impossible to access quality abortion care, which can have numerous harmful consequences for women's health and lives and undermine their human rights.<sup>51</sup>

The Guideline outlines a series of recommendations for member states on good practice in the areas of law and policy, clinical services, and service delivery. The recommendations are based on human rights law and public health evidence and, together with the best practices described in the Guideline, they aim to enable evidence-based decision making with respect to quality abortion care.<sup>52</sup>

The recommendations include the following and are intended to:

- ✓ Reform laws to fully decriminalise abortion.<sup>53</sup>
- Legalise abortion on request and remove requirements that mandate third party authorisation by any individual, body, or institution.<sup>54</sup>
- ✓ Repeal laws and regulations that restrict abortion by grounds, prohibit abortion based on gestational limits, and require mandatory waiting periods.<sup>55</sup>
- ✓ Ensure access to medication abortion including the option of selfmanagement without the direct supervision of a trained health worker and the option of telemedicine in early pregnancy.<sup>56</sup>
- ✓ End regulations that limit who can provide and manage abortion care that are inconsistent with WHO guidance.<sup>57</sup>
- ✓ Guarantee that access to and continuity of comprehensive abortion care be protected against barriers created by health workers' refusals of care.<sup>58</sup>

#### 3.2 Highly restrictive abortion laws

Currently five Council of Europe member states – Andorra, Liechtenstein, Malta, Monaco, and Poland – retain highly restrictive abortion laws.<sup>59</sup> These laws have myriad harmful consequences for women and girls and other persons in these countries who need access to abortion care. By restricting the legality of abortion to specific exceptions or grounds, such laws exclude most people who need abortion care from the possibility of legal access. As a result, they are forced to seek abortion care outside of legal pathways or to travel to other jurisdictions, if they have the means to do so. For those who are unable to access abortion in their home country or travel to other countries, they may be forced to continue a pregnancy against their will.<sup>60</sup>

#### 3.3 Criminalisation of abortion

Although most member states have legalised abortion on request, in most cases abortion is still regulated in part through criminal law. In several cases – such as in Belgium, Cyprus, Iceland, Italy, Liechtenstein, Portugal, and San Marino – these laws criminalise women who undergo abortion outside the scope of the law.<sup>61</sup> However, more commonly, member states' laws criminalise the actions of health care providers or others who assist people to access abortions that are not lawful.

Such laws treat abortion care differently from all other forms of health care and can cause significant harm to the health and wellbeing of those

seeking abortion care. They also exert a chilling effect on health care workers' provision of legal abortion care, which can delay access to care and lead to denial of life-saving treatment.<sup>62</sup> Women denied abortion care due to criminalisation have reported experiencing severe physical and mental anguish, extreme vulnerability, stigma, and isolation.<sup>63</sup>

For example, in Poland, the tightening of already highly restrictive law through a regressive ruling of the country's Constitutional Tribunal, issued in 2020, has resulted in denials of life-saving treatment to several women in obstetric emergencies. It has been accompanied by an increase in the number of criminal investigation proceedings initiated against persons suspected of assisting with access to abortion care outside the narrow scope of the law.<sup>64</sup>

#### 3.4 Access barriers

In most member states, a range of regulatory, policy and practical barriers continue to hinder access to abortion care in practice. These include medically unnecessary restrictions such as short time limits for abortion on request, third-party authorisation requirements, restrictions on access to medication abortion, and restrictions on who can provide and manage abortion that are inconsistent with WHO guidance. In several countries – namely Albania, Armenia, Belgium, Georgia, Germany, Hungary, Ireland, Italy, Latvia, Luxembourg, Portugal, and Slovakia – abortion laws impose mandatory waiting periods. Some countries, including Albania, Armenia, Georgia, Germany, Hungary, Italy, Lithuania, and Slovakia maintain mandatory counselling or information requirements that are sometimes intended to dissuade women from accessing abortion care. For instance, laws in Germany and Hungary require directive counselling explicitly intended to influence decision making.65 In addition, some member states - such as Croatia, Czechia, Lithuania, Romania and Slovakia - still do not include abortion on request in public health insurance schemes, meaning that those seeking abortion on request have to pay for it out of their own pockets.66 Lack of competency-based training for health workers on the provision of quality abortion care is also of concern in several member states.

These barriers delay access to abortion care and may force women to seek abortion care outside of legal frameworks, travel to another country for care, or continue a pregnancy against their will. These barriers disproportionately impact marginalised groups, such as women living in poverty or facing financial hardship, rural women, and adolescent girls, who may not always be able to comply with these regulatory requirements.<sup>67</sup>

In addition, many member states still have not established adequate mechanisms to ensure that refusals of abortion care on the grounds of conscience or religion by medical providers do not hinder timely access to care. In several places, laws and policies regulating refusals of care are inadequate, and effective monitoring and oversight mechanisms to ensure compliance are lacking. As a result, in Croatia, Italy, Northern Ireland, Poland, Romania, and Slovakia, for example, widespread refusals of care are imposing significant burdens on those who need abortion care and are leading to delays in care, thereby endangering the health and life of patients.<sup>68</sup> The authorities' failure to take effective measures to ensure service provision also impacts the health and wellbeing of medical professionals who do provide abortion care and can sometimes lead to career limitations and discrimination.<sup>69</sup> Similarly, the failures of some member states to ensure those seeking and providing abortion care are effectively protected from harassment and intimidation also continue to be of concern, as is abortion stigma that remains pervasive across the region.<sup>70</sup>

#### 3.5 Access to evidence-based information

Although some improvement has been made in the availability of evidence-based information on abortion and its regulation, easy access to such information continues to be a challenge in some member states. Only a small number of countries provide information on abortion care on government-run websites<sup>71</sup> and some member states still maintain laws that prohibit public dissemination of some or all forms of evidence-based information.<sup>72</sup>

Furthermore, websites providing false and misleading information about abortion designed to dissuade people from accessing abortion care have spread across Europe and new organisations have been established with the purpose of providing misleading and biased information and counselling on abortion.<sup>73</sup> There have also been attempts to mandate the provision of biased information about abortion by health professionals and to ban the public provision of evidence-based information on abortion.<sup>74</sup>

#### 3.6 Retrogression

In a small number of member states, highly concerning efforts to roll back entitlements to abortion have advanced in recent years, as policy makers have pursued legislative and judicial initiatives seeking to remove grounds for legal abortion and introduce new access barriers.<sup>75</sup>

The most egregious example of retrogression took place in Poland between 2020 and 2021 when legal provisions allowing abortion in situations of severe or fatal foetal impairment were repealed following a regressive ruling by the Constitutional Tribunal.<sup>76</sup> Since then, the already severe impact of the country's highly restrictive abortion law has increased even further, leading many women to experience anxieties about becoming pregnant.<sup>77</sup> Several women have died in Polish hospitals in recent years, reportedly because they were denied life-saving care during obstetric emergencies in connection with the chilling effect of the retrogressive restrictions on abortion access.<sup>78</sup> In December 2023, the European Court of Human Rights held that rollbacks on legal entitlements to abortion, following the ruling of the Constitutional Tribunal involving judges appointed by a severely irregular procedure that did not meet rule of law requirements, violated a woman's right to respect for private life under the European Convention on Human Rights.<sup>79</sup>

Retrogression also occurred in Hungary in 2022, when new regulations were introduced requiring patients to listen to the foetal ultrasound prior to obtaining abortion care.<sup>80</sup> In 2023, Georgia also adopted further restrictive measures, including the requirement to include a psychologist and a social worker in the provision of mandatory counselling before abortion care.<sup>81</sup> Attempts to restrict access to abortion have also taken place in other member states in recent years – notably in Italy, Lithuania, and Slovakia.<sup>82</sup>

Retrogression that leads to the removal of pre-existing entitlements to abortion or the introduction of new restrictions on access to abortion care is contrary to international human rights law.<sup>83</sup> Human rights mechanisms have repeatedly urged states to refrain from adopting retrogressive proposals seeking to restrict or undermine access to abortion care.<sup>84</sup>

# Chapter 4 **Obstetric and gynaecological care**

In recent years, some Council of Europe member states have made advancements towards ensuring standards of care and access to quality reproductive health care. Yet, important public health, human rights, equality, and structural concerns persist across Europe that continue to undermine equal access to quality obstetric and gynaecological care. Research into the experiences of women in reproductive health care settings, particularly during facility-based childbirth, has highlighted instances of disrespectful, abusive, and discriminatory treatment across Council of Europe member states. The COVID-19 pandemic and associated measures compromised the provision of quality obstetric care in some settings as restrictions were placed on the presence of birth companions and visitors, mandatory caesarean delivery was imposed for those who tested positive with COVID-19, and newborns were often separated following birth.<sup>85</sup> In some member states, the legacy of these measures persists and pandemic-related restrictions have not been fully removed in obstetric care settings.

## 4.1 Developments in international human rights standards and public health guidelines

International human rights mechanisms have underlined that disrespectful, abusive, and discriminatory treatment during obstetric care and in other reproductive health services gives rise to violations of human rights and they have called for reforms to ensure respect for free and informed consent and the provision of respectful care. In recent years, they have increasingly addressed the content of states' obligations to guarantee respectful care during pregnancy and childbirth, free of discrimination, coercion, and violence.86

For example, the United Nations Special Rapporteur on Violence against Women, its Causes and Consequences has underlined that violations of

informed consent, restrictions on decision making, and disrespect and mistreatment in reproductive health services and during childbirth is systematic in nature and is part of "a continuum of the violations that occur in the wider context of structural inequality, discrimination and patriarchy, and are also the result of a lack of proper education and training, as well as a lack of respect for women's equal status and human rights."87 The Special Rapporteur specified that mistreatment and violence against women in obstetric care and other reproductive health care settings results in physical and psychological suffering that violates numerous human rights, including the rights to freedom from torture and ill-treatment, life, health, privacy, information and freedom from discrimination.88 She also recognised that the conditions and constraints of the health system and power dynamics in the provider-patient relationship are the root causes of mistreatment and violence against women during childbirth, and that these constraints are aggravated by discriminatory laws or practices and harmful gender stereotypes regarding women's decision-making competence, their role in society and motherhood.89 The Special Rapporteur has urged states to respect, protect and fulfil women's human rights during reproductive health care services and childbirth, free from mistreatment and genderbased violence, and ensure women receive dignified and respectful reproductive health care and obstetric care, free from discrimination and violence, and allocate sufficient budgetary resources needed to provide quality, accessible reproductive and maternal health care.90

Similarly, in three recent cases, the United Nations Committee on the Elimination of Discrimination against Women found that violations of women's human rights occurred in situations where pregnant women were subjected to multiple vaginal examinations, where episiotomies or caesarean sections were carried out without their informed consent, and where their consent was not sought for feeding choices for newborns.<sup>91</sup> The Committee found that obstetric care must be provided in a way that ensures informed consent, respects dignity, guarantees confidentiality and is sensitive to the pregnant women's needs and perspectives.<sup>92</sup> It outlined that pregnant women "have the right to receive full information about recommended treatments so that they can make well-considered and informed decisions" and that medical professionals must be provided with professional training on women's human rights.<sup>93</sup>

Human rights mechanisms have also emphasised the importance of ensuring the individual's free and informed consent for other reproductive health care procedures and interventions.<sup>94</sup> They have found rights violations where medical procedures such as abortion or intra-uterine contraception were performed without the patient's free and informed consent.<sup>95</sup>

Updated public health guidelines have also been issued setting out guidance for member states. For example, in 2018, the World Health Organization published comprehensive and consolidated guidelines on intrapartum care for a positive childbirth experience, highlighting the importance of woman-centred care to optimise the experience of labour and childbirth for women and their babies through a holistic, human rights-based approach.<sup>96</sup>

#### 4.2 Disrespectful care and mistreatment

In many member states, failures to ensure adequate standards of care and respect for the dignity and autonomy of women in obstetric care persist. Women across the region report experiences of mistreatment, humiliation, verbal, psychological and physical abuse, and neglect in the context of facility-based childbirth. These include restrictions on movement, fundal pressure, denial of adequate pain relief, condescension, scolding, mocking and other forms of verbal disrespect.<sup>97</sup> Experiences of neglect, delays, denial of care, inadequate pain relief, sexual violence, humiliation, verbal abuse, judgmental and sexist remarks, and violations of privacy have also been reported in gynaecological care settings.<sup>98</sup> Although these practices frequently affect women from all backgrounds, reports indicate that women from marginalised groups including women living in poverty, migrants, women from ethnic minorities, as well as other persons who might require relevant care, such as non-binary people and trans men, face exacerbated harm.<sup>99</sup>

The impact of such disrespectful and abusive practices on women's emotional and mental health and wellbeing can be severe. Abusive and disrespectful obstetric care has been associated, for example, with postpartum depression, post-traumatic stress disorder, and negative implications for sexuality and sexual expression. It has also been associated with exacerbated risks of complications during childbirth and decreased trust in the health system leading to increased unwillingness to seek medical care.<sup>100</sup>

These practices are often connected with structural problems and systemic conditions that significantly impact the provision of obstetric care and other reproductive health services, including the lack of adequate health system investment, staff and equipment shortages, poor working conditions for health care staff, and lack of adequate and clear policies, guidelines, and training.<sup>101</sup> Harmful gender stereotypes, stigma, and discriminatory attitudes also play a role in shaping the treatment of women in obstetric care and other reproductive health care settings.

#### 4.3 Informed consent and decision making

Allegations of disregard for women's autonomy and decision-making capacity during childbirth, and related failures to ensure women's full and informed consent prior to medical interventions, continue to be common in obstetric care settings in many member states. Women report a lack of clear and complete information about the decisions made by medical staff during childbirth and report that sometimes procedures are performed despite their explicit disapproval and lack of consent, such as episiotomies and vaginal examinations. Women also indicate that sometimes procedures are performed in the presence of students without their consent.<sup>102</sup>

Additionally, concerns regarding involuntary sterilisation, contraception and abortion, as well as other medical procedures performed without the free and informed consent of women with disabilities persist across the region, despite the fact that such acts are specifically prohibited under the Council of Europe Convention on preventing and combating violence against women and domestic violence, often referred to as the 'Istanbul Convention' and the Convention on the Rights of Persons with Disabilities. 103

### Chapter 5

# Access to sexual and reproductive health and rights without discrimination

In all member states, intersectional discrimination continues to impede the full realisation of sexual and reproductive health and rights of marginalised groups who face the brunt of ongoing shortfalls, restrictions, and inequalities in sexual and reproductive health and rights. Multiple and intersecting inequalities in the enjoyment of the social determinants of health - including income security and social protection, adequate living conditions, education and employment conditions - also affect sexual and reproductive health equity and outcomes. National level data on sexual and reproductive health indicators often mask extreme disparities in health outcomes amongst different groups of people within a country, and marginalised groups of women and girls are still regularly excluded from policy development and evaluation processes.

## 5.1 Developments in international human rights standards

Under international human rights law, states are obliged to eliminate all forms of discrimination in the enjoyment of sexual and reproductive health and rights. <sup>105</sup> In recent years, human rights mechanisms have provided more guidance to member states on the content of this obligation, by urging them to address and combat multiple and intersecting forms of discrimination and to mitigate the aggravating effects of such discrimination on the enjoyment of sexual and reproductive health and rights through legislation, administrative, policy changes and special measures. <sup>106</sup>

For instance, they have elaborated on member states' obligations to ensure that women with disabilities can fully enjoy their sexual and reproductive health and rights, by taking effective measures to enable women with disabilities to make autonomous decisions about their sexual and

reproductive health and to ensure their access to sexual and reproductive health services, including abortion care, free from coercion and any form of discrimination.<sup>107</sup> They have also urged member states to eliminate the practice of segregation of Roma women in reproductive health care settings, to bring an end to different forms of racist verbal, physical and psychological violence that Roma women often face in these health care contexts, and to ensure non-discriminatory and adequate access to sexual and reproductive health care for Roma women.<sup>108</sup> They have also called on states to ensure equal access to sexual and reproductive services for LGBTI persons and to address stigma they experience when seeking these services. 109 Human rights mechanisms have also underlined that member states are required to ensure that all refugees, asylum seekers and migrants, particularly those in an irregular situation, and stateless persons can access affordable quality sexual and reproductive health care, regardless of their migration status<sup>110</sup> and they have also specified that pregnant and breastfeeding women should not be subjected to administrative detention.<sup>111</sup> In addition, they have emphasised the importance of guaranteeing access to comprehensive sexual and reproductive health care in conflict- and crisis- settings and mitigating the impact of climate change on the realisation of sexual and reproductive health and rights.<sup>112</sup>

#### 5.2 Challenges and deficits

Across the region, marginalised groups continue to face discrimination when it comes to the enjoyment of their sexual and reproductive health and rights. Specific challenges are faced by, amongst others, women and girls living in poverty, rural women and girls, women and girls with disabilities, ethnic minorities including Roma women and girls, adolescents, older women, unmarried and single women, women living with HIV, sex workers, women and girls affected by conflict situations, victims of human trafficking, refugees, asylum seekers, migrant women in an irregular situation, and LGBTI people.<sup>113</sup>

The Council of Europe Commissioner for Human Rights has repeatedly called on Council of Europe member states to guarantee the right to health, including the right to sexual and reproductive health, of persons belonging to specific groups facing intersecting forms of discrimination, including women with disabilities, Roma women, migrant women and LGBTI people.<sup>114</sup>

By way of example, in much of the region **Roma women and girls** continue to face deeply entrenched racism and discrimination, including in relation to the social determinants of health, which undermines their sexual and reproductive health and rights.<sup>115</sup> In some member states,

they are subjected to segregation, racial harassment and abuse in reproductive health care settings. 116 In some cases, Roma women survivors of forced sterilisations continue to be deprived of effective remedies, and comprehensive mechanisms to address their situation are still needed. 117 In other cases, mechanisms have been developed that aim to provide survivors of forced sterilisation with reparations for the harm they endured, although problems in implementation raise questions about the extent to which these are currently able to provide for effective remedies. 118 Roma women who have other intersecting identities often face huge barriers as well as forms of discrimination in the access and provision of sexual and reproductive health care services. 119

Similarly, in most member states, **women and girls with disabilities** are often subjected to serious violations of their sexual and reproductive health and rights.<sup>120</sup> Many women and girls with disabilities undergo involuntary sterilisation, contraception and abortion without their free and informed consent.<sup>121</sup> Substitute decision-making systems, where a guardian or a judge is empowered to make life-altering decisions for persons with disabilities, against their will and preferences, are common in many European countries.<sup>122</sup> Women with disabilities also face huge barriers in access to sexual and reproductive health care and evidence-based information.<sup>123</sup>

The provision of appropriate sexual and reproductive health care services and information for **older women** is also lacking in some countries. Older women face increased risks of violence, abuse, and neglect as well as harmful gender stereotypes, assumptions, and stigma in relation to sexuality and sexual and reproductive health. Health care policy and service provision often fails to prioritise the specific sexual needs of women related to, and following, the menopause, and age-related financial barriers often restrict older women's access to comprehensive quality care. Older women in care settings are especially vulnerable to sexual violence or other serious violations of their personal and bodily integrity.<sup>124</sup>

Across Europe, refugees, asylum seeking and **migrant women**, especially women in an irregular situation, also face serious forms of discrimination and exclusion in relation to their enjoyment of sexual and reproductive health and rights. In many member states, laws and policies directly or indirectly prevent migrant women in an irregular situation, refugees, and asylum seekers from accessing affordable sexual and reproductive health care, including during pregnancy and childbirth.<sup>125</sup> For many migrant women, being in an irregular situation, fear of being reported to immigration authorities, and a lack of effective separation — or "firewalls" — between health services and immigration controls and enforcement prevents them

from accessing sexual and reproductive health care, sometimes placing their health and lives at risk.<sup>126</sup> In administrative detention settings or transit zones, women can face barriers accessing adequate and appropriate sexual and reproductive health care.<sup>127</sup>

Although there have been important positive developments in many member states, **LGBTI people** continue to experience diverse forms of intersectional discrimination in the sphere of sexual and reproductive health and rights. <sup>128</sup> In some countries, discrimination on the basis of marital status, health status, sex, sexual orientation, gender identity and expression, as well as sex characteristics, continues to limit the rights of LGBTI people to access essential sexual and reproductive health services. In some parts of the region, LGBTI people still face practices that violate their sexual and reproductive health and rights, such as so-called "conversion therapies", <sup>129</sup> sterilisation of trans people, <sup>130</sup> medically unnecessary "normalising" treatments on intersex children without their prior, free, and informed consent <sup>131</sup> and discriminatory restrictions on access to assisted reproduction. <sup>132</sup>

**Sex workers** across Europe also face significant obstacles to securing equal and unhindered access to their sexual and reproductive health and rights.<sup>133</sup> Stigma and discrimination towards them in health care settings is particularly widespread. This may result in denial of care, disrespectful and abusive language and treatment, confidentiality breaches and a lower quality of care, and some sex workers avoid seeking health care altogether. They sometimes lack access to social protection, health insurance and health care subsidisation schemes. Barriers are often exacerbated for sex workers who face intersecting forms of discrimination, including on the basis of age, disability, sexual orientation, gender identity and expression, sex characteristics, migration status, and health status.<sup>134</sup>

### Chapter 6

# Human rights defenders working to advance sexual and reproductive health and rights

Continued progress in sexual and reproductive health and rights in Europe would not be possible without the tireless work, resolve, and courage of the human rights defenders and civil society organisations that are working across the region to advance these rights. Council of Europe member states have clear international human rights obligations to protect human rights defenders and create an enabling environment in which they can conduct their work, free from hindrance and insecurity.<sup>135</sup>

However, despite these obligations, human rights defenders and civil society organisations working to promote and protect sexual and reproductive health and rights in some parts of Europe face various concerning challenges and risks. In some member states, recent rollbacks on sexual and reproductive rights and gender equality, alongside efforts to undermine the rule of law, have exacerbated these threats. Some human rights defenders are facing threats to their personal safety and liberty, as well as different forms of intimidation and harassment. In some countries, civil society organisations working on sexual and reproductive health and rights face severe financial and operational uncertainty and insecurity.<sup>136</sup> In some contexts, medical professionals who provide sexual and reproductive health care are also facing similar threats and challenges.<sup>137</sup>

Human rights mechanisms have expressed increasing concern about the threats and challenges faced by human rights defenders and organisations working in Europe. They have recognised that women human rights defenders and women's rights organisations, that are often at the front lines of providing direct support and assistance to those in need of sexual and reproductive health care, may be exposed to gender-specific forms of persecution and discrimination because of their human rights work. They have urged states to take effective measures to ensure safe and favourable environments for human rights defenders working on sexual

and reproductive health and rights and to support them in discharging their role.<sup>138</sup>

## 6.1 Risks to personal safety and fear of prosecution and judicial harassment

In some Council of Europe member states, human rights defenders report receiving threats to their personal safety because of their advocacy and efforts to promote and protect sexual and reproductive rights.

They have reported experiencing incidents and threats of physical attack and violence, including sexual violence and death threats, stigmatisation, smear campaigns, hate speech, defamation, and harassment, both online and offline.<sup>139</sup>

In some countries, human rights defenders have also faced legal consequences or harassment because of their work. 140 For example, in 2023, Justyna Wydrzyńska was convicted by a district court in Poland for helping a woman access abortion medication. 141 In Andorra, Vanessa Mendoza Cortés faced criminal charges, including on counts of a "crime against the prestige of the institutions", for her work in 2019 voicing concern to the United Nations Committee on the Elimination of Discrimination against Women and in the media about women's human rights and Andorra's abortion ban. In January 2024, she was acquitted by the Tribunal de Corts in Andorra. 142

Human rights defenders who have exercised their right to freedom of assembly are also facing legal consequences. For example, in Poland, peaceful demonstrators who gathered to protest against rollbacks on abortion rights reported being subjected to intimidation and excessive use of force by law enforcement forces and some of them faced criminal charges related to their role in these demonstrations.<sup>143</sup>

The use of criminal investigation and prosecution as a form of reprisal against human rights defenders for their work to protect human rights not only causes substantial harm to the individuals concerned but can also have a broader chilling effect. In some member states, human rights defenders working to advance sexual and reproductive health and rights report increasing levels of fear that they will be targeted through legal channels by the state authorities.

#### **6.2 Unfavourable and hostile working environments**

Many human rights defenders and civil society organisations working to advance sexual and reproductive rights in Europe also face challenges related to their working environment. In several member states, there are concerns about increasing restrictions on civil society space and clampdowns on the legitimate activities of human rights defenders and civil society organisations through legislative attempts to place unjustified restrictions on the activities and funding sources of civil society organisations.<sup>144</sup>

A sustained lack of funding, including sources of state support, is another challenge faced by many civil society organisations working in Europe to advance sexual and reproductive health and rights. Long-term financial insecurity has negatively impacted organisations' ability to respond to retrogression and to proactively protect sexual and reproductive health and rights. In some member states, there are concerning reports of state funding being diverted to anti-SRHR (anti-Sexual and Reproductive Health and Rights) or anti-LGBTI organisations.<sup>145</sup>

A rise in stigmatising political rhetoric and attacks on sexual and reproductive health and rights and gender equality also compounds the challenges faced by human rights defenders and civil society organisations working in this field. The ability of human rights defenders to continue their work towards advancing sexual and reproductive rights, women's rights and gender equality is often undermined as they must dedicate considerable time and resources towards halting initiatives that seek to rollback sexual and reproductive health and rights.<sup>146</sup>

# The Commissioner's recommendations

The Commissioner for Human Rights calls on Council of Europe member states to fully implement their human rights obligations in the field of sexual and reproductive health and rights.

The recommendations outlined below complement the comprehensive recommendations to member states set out by the Commissioner in the 2017 Issue Paper on *Women's sexual and reproductive health and rights in Europe* and provide guidance for member states on actions required of them to guarantee compliance with international human rights law and standards.

The Commissioner for Human Rights recommends that Council of Europe member states:

### Guarantee the provision of comprehensive sexuality education:

- » implement and mainstream mandatory comprehensive sexuality education (CSE) across the education system, including into ordinary school curricula, and ensure that CSE is age-appropriate and developmentally appropriate and provided in line with international human rights law and public health guidelines
- » ensure the comprehensive nature of sexuality education curricula, including topics related to sexual and reproductive autonomy, gender equality, sexual orientation and gender identity, gender-based violence, consent, diversity, and healthy and respectful relationships
- » reform laws and policies to ensure the mandatory nature of age appropriate CSE in school curricula and consult children and young people in the design, implementation, and evaluation of CSE programming
- » provide teachers and education professionals with sufficient specialised training and support on an ongoing basis and guarantee adequate financial and human resources to allow for the delivery of high-quality CSE

» establish CSE programmes for out-of-school children and young people

Guarantee the availability and accessibility of affordable contraceptive services and evidence-based information on contraception:

- » ensure the affordability of the full range of modern contraceptive methods and cover them under public health insurance or subsidisation schemes, as well as ensure that coverage extends to all people of reproductive age and all methods of modern contraception
- » include all modern contraceptive methods in national lists of essential medicines and ensure their practical availability across rural and urban areas
- » disseminate evidence-based, accurate information on contraception, in accessible formats and languages, and establish awareness-raising programmes and strategies to tackle and dispel misconceptions
- » repeal residual legal and policy barriers that impede access to contraceptive services and information, such as prescription requirements for emergency contraception or third-party authorisation requirements, including for adolescents and women with disabilities
- » provide regular, specialised and evidence-based training for health care workers on the delivery of contraceptive education, information and services

Guarantee access to quality abortion care including evidencebased information on abortion:

- » repeal laws and policies that criminalise abortion, so as to remove all criminal penalties for consensual abortion, including for anyone who undergoes abortion, assists with access to abortion or provides abortion care
- » repeal highly restrictive abortion laws and legalise abortion on request, as well as remove all regulatory barriers for access to abortion care in line with international human rights standards and the World Health Organisation abortion care guidelines
- » disseminate accessible and evidence-based information on abortion care and related legal entitlements and reform laws and policies to ensure that abortion counselling is never mandatory, biased, or directive

- » ensure the availability of medication abortion and allow selfmanagement of abortion medication in accordance with the World Health Organization guidelines
- » guarantee timely access to abortion care in practice, including by ensuring that:
  - it is provided free of charge or covered under public health insurance
  - there are enough fully trained providers of abortion care throughout the country
  - access to abortion is not obstructed by refusal of care on the part of health care workers
  - measures are in place to protect from harassment, intimidation, and stigmatisation those seeking and providing care

## Ensure dignified and respectful gynaecological and obstetric care, free from discrimination and violence:

- » ensure that all forms of gynaecological and obstetric care are provided in line with human rights standards, including the principle of nondiscrimination and respect for the dignity, autonomy, integrity, and decision-making capacity of those requiring gynaecological and obstetric care
- » ensure that full and effective informed consent policies and protocols are applied to all forms of reproductive health care, including obstetric care
- » repeal laws and policies that directly or indirectly restrict individuals' decision making in relation to their sexual and reproductive health, and ensure that everyone can benefit from the presence of a skilled birth attendant and a companion of their choice during childbirth, without discrimination on any grounds
- » eradicate medically unnecessary and harmful practices that undermine autonomy and decision making during and postchildbirth
- » conduct information and awareness-raising campaigns on patients' rights, including on preventing and combating sexism and genderbased violence in gynaecological and obstetric care
- » prohibit and sanction disrespect, mistreatment, and abuse in gynaecological and obstetric care

Establish effective measures to tackle intersecting forms of discrimination that limit the equal enjoyment of sexual and reproductive health and rights:

- » repeal discriminatory laws and policies, and end discriminatory practices that curtail access to sexual and reproductive health care for certain groups, including on the grounds of age, disability, ethnicity, gender identity and expression, nationality or migration status, relationship status, sex, sex characteristics, sexual orientation, and socio-economic status
- » establish effective public information and awareness-raising campaigns and training programmes designed to eradicate multiple and intersectional discrimination in the sphere of sexual and reproductive health and rights, particularly in health care settings
- » ensure that persons with disabilities are not denied their right to autonomy in respect of their sexual and reproductive health care and for this purpose ensure that no medical intervention is performed without their prior, free, and informed consent
- » ensure a life course approach in the provision of sexual and reproductive health services, from early childhood and throughout old age, and combat ageism within health care settings
- » eradicate practices of segregation on the grounds of ethnicity in sexual and reproductive health care settings and ensure that Roma women can access quality care, free from discrimination, disrespect, and abuse
- » ensure that all survivors of sexual violence, including those in conflict zones, victims of human trafficking, asylum seekers and refugees, and migrants in administrative detention can access comprehensive sexual and reproductive health services, including emergency contraception, abortion care and HIV post-exposure prophylaxis
- » guarantee sex workers' rights to be free from violence, stigma, and discrimination in health care settings, as well as to quality sexual and reproductive health care

Ensure there is no retrogression in the protection of sexual and reproductive health and rights and enable the work of human rights defenders acting to advance sexual and reproductive health and rights:

- » prevent the erosion of existing protections in the field of sexual and reproductive health and rights; reject measures and initiatives that seek to roll back established entitlements and repeal any retrogressive measures
- » refrain from rhetoric and discourse that are contrary to human rights principles and that undermine commitments to women's rights, gender equality and sexual and reproductive health and rights or that stigmatise and delegitimise human rights defenders working in this field
- » promote men's and boys' engagement and accountability in matters relating to sexism, misogyny, violence, and toxic masculinity, as well as sexual and reproductive health and rights of women and girls in all their diversity
- » reform laws and policies that undermine the operation of human rights defenders, civil society organisations, and health care providers acting to advance sexual and reproductive health and rights, whilst also eradicating, preventing and sanctioning violence, hate speech, smear campaigns, harassment, and intimidation which target these actors
- » ensure that civil society organisations working to advance sexual and reproductive health and rights have access to public funds in a non-discriminatory and transparent manner and that they effectively enjoy the freedom to solicit and receive funds from institutional or individual donors in other countries
- » engage systematically human rights defenders and civil society organisations that are working to advance sexual and reproductive health and rights in the development, implementation, monitoring, evaluation and continuing improvement of law, policy and practice relating to sexual and reproductive health and rights

#### **Endnotes**

- 1. Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual and reproductive health and rights in Europe* (2017), available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead.
- 2. UNESCO, UNICEF, UNAIDS, UNFPA, WHO, UN Women, *The journey towards comprehensive sexuality education Global status report* (2021) p.6, available at https://unesdoc.unesco.org/ark:/48223/pf0000379607; UNESCO, UNAIDS, UNFPA, UNICEF, UN WOMEN, WHO, *International technical guidance on sexuality education: an evidence-informed approach* (2018), available at https://unesdoc.unesco.org/ark:/48223/pf0000260770.
- 3. UNESCO, UNAIDS, UNFPA, UNICEF, UN WOMEN, WHO, *International technical guidance on sexuality education: an evidence-informed approach* (2018), available at https://unesdoc.unesco.org/ark:/48223/pf0000260770.
- 4. Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual* and reproductive health and rights in Europe, (2017), pp. 57-58, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead; Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health et al., *A Compendium* on Comprehensive Sexual Education (2023), available at https://www.ohchr.org/sites/default/files/documents/issues/health/sr/Compendium-Comprehensive-Sexuality-Education-March-2023.pdf.
- 5. See, e.g., Human Rights Committee (HRC), General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8; CEDAW, Report of the inquiry concerning the United Kingdom of Great Britain and Northern Ireland under Article 8 of the Convention on the Elimination of All Forms of Discrimination against Women, CEDAW/C/OP.8/GBR/1, (2018), para. 75; Committee on Economic, Social and Cultural Rights (CESCR), Concluding Observations: Denmark, E/C.12/DNK/CO/6 (2019), para. 61; Council of Europe Commissioner for Human Rights, Comprehensive sexuality education protects children and helps build a safer, inclusive society, Human Rights Comment published on 21 July 2020, available at https://www.coe.int/en/web/commissioner/-/comprehensive-sexuality-education-protects-children-and-helps-build-a-safer-inclusive-society.
- 6. Committee on the Rights of Persons with Disabilities (CRPD), Concluding Observations: Switzerland, CRPD/C/CHE/CO/1 (2022), para. 50(e); Committee on the Elimination of Discrimination against Women (CEDAW), Concluding Observations: Bulgaria, CEDAW/C/BGR/CO/8 (2020), para. 34(b); CESCR, Concluding Observations: Azerbaijan, E/C.12/AZE/CO/4 (2021), para. 47.
- 7. See, e.g., Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health et al., *A Compendium on Comprehensive Sexual Education* (2023), available at https://www.ohchr.org/sites/default/files/documents/issues/health/sr/Compendium-Comprehensive-Sexuality-Education-March-2023.pdf; Committee on the Rights of the Child (CRC), Concluding Observations: Sweden, CRC/C/SWE/CO/6-7 (2023), para. 34(a); Concluding Observations:

- Ireland, CRC/C/IRL/CO/5-6 (2023), para. 33(b); Concluding Observations: Kingdom of the Netherlands, CRC/C/NLD/CO/5-6 (2022), para. 30(b); Concluding Observations: Hungary, CRC/C/HUN/CO/6 (2020), para. 33(c); CEDAW, Concluding Observations: Georgia, CEDAW/C/GEO/CO/6 (2023), para. 34(b); Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), para. 33(a); Concluding Observations: Azerbaijan, CEDAW/C/AZE/CO/6 (2022), paras. 30(c), 34(g); Concluding Observations: Bulgaria, CEDAW/C/BGR/CO/8 (2020), para. 34(b).
- 8. CRC, General comment No. 25 (2021) on children's rights in relation to the digital environment, CRC/C/GC/25 (2021), paras. 104-105.
- 9. See, e.g., CRC, Concluding Observations: United Kingdom of Great Britain and Northern Ireland, CRC/C/GBR/CO/6- 7 (2023), para. 44(b); CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), para. 42(f); CEDAW, Concluding Observations: Iceland, CEDAW/C/ISL/CO/9 (2023), para. 32(a); Concluding Observations: Slovenia, CEDAW/C/SVN/CO/7 (2023), para. 36(c); Concluding Observations: Türkiye, CEDAW/C/TUR/CO/8 (2022), para. 44(f); Concluding Observations: Belgium, CEDAW/C/BEL/CO/8 (2022), para. 42(c); Concluding Observations: Denmark, CEDAW/C/DNK/CO/9 (2021), para. 35(c); Concluding Observations: Latvia, CEDAW/C/LVA/CO/4-7 (2020), para. 38(d); Council of Europe Commissioner for Human Rights, *Comprehensive sexuality education protects children and helps build a safer, inclusive society*, Human Rights Comment published on 21 July 2020, available at https://www.coe.int/en/web/commissioner/-/comprehensive-sexuality-education-protects-children-and-helps-build-a-safer-inclusive-society.
- 10. UNESCO, UNAIDS, UNFPA, UNICEF, UN WOMEN, WHO, *International technical guidance on sexuality education: an evidence-informed approach* (2018, available at https://unesdoc.unesco.org/ark:/48223/pf0000260770.
- 11. Ibid., p. 16.
- 12. German Federal Centre for Health Education (BZgA), International Planned Parenthood Federation European Network (IPPF EN), Sexuality Education in Europe and Central Asia: State of the Art and Recent Developments; An Overview of 25 Countries. Assessment Report, (2018), available at https://www.bzga-whocc.de/en/publications/studies/; Policy Department for Citizens' Rights and Constitutional Affairs, Comprehensive sexuality education: why is it important? (2022), Study requested by the FEMM committee, available at https://www.europarl.europa.eu/RegData/etudes/STUD/2022/719998/IPOL\_STU(2022)719998\_EN.pdf.
- 13. Policy Department for Citizens' Rights and Constitutional Affairs, *Comprehensive sexuality education: why is it important?* (2022), Study requested by the FEMM committee, available at https://www.europarl.europa.eu/RegData/etudes/STUD/2022/719998/IPOL\_STU(2022)719998\_EN.pdf.
- 14. Ibid., pp. 38-39.
- 15. See, e.g., Women's and girls' sexual and reproductive health rights in crisis, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), paras. 46, 79(e); Council of Europe, Recommendation CM/Rec(2019)1 on preventing and combating sexism, adopted by the Committee of Ministers on 27 March 2019; The place of men and boys in gender equality policies and in policies to combat violence against women, Guidelines CM(2023)51-add2-final, adopted by the Committee of Ministers on 5 June 2023, , available at https://rm.coe.int/prems-111923-gbr-2573-guideline-cmrec-2023-51-text-bat-a5-web/1680ac7900.
- 16. Ibid., page 54-55.
- 17. BZgA, IPPF EN, Regional Overview of the Status of Sexuality Education in 25 Countries of the WHO European Region (2018), available at https://www.bzga-whocc.de/en/publications/studies/.

- 18. Policy Department for Citizens' Rights and Constitutional Affairs, *Comprehensive sexuality education: why is it important?* (2022), Study requested by the FEMM committee, pp. 39, 52, available at https://www.europarl.europa.eu/RegData/etudes/STUD/2022/719998/IPOL\_STU(2022)719998\_EN.pdf.
- 19. See ibid., pp. 37, 44-45.
- 20. Council of Europe Commissioner for Human Rights, *Comprehensive sexuality education protects children and helps build a safer, inclusive society*, Human Rights Comment published on 21 July 2020, available at https://www.coe.int/en/web/commissioner/-/comprehensive-sexuality-education-protects-children-and-helps-build-a-safer-inclusive-society.
- 21. Hungary: Amendments affecting LGBTQI people incompatible with international human rights standards, according to Venice Commission, Media Release (14.12.2021), available at https://search.coe.int/directorate\_of\_communications/Pages/result\_details. aspx?ObjectId=0900001680a4dd13; Hungary passes law banning LGBT content in schools or kids' TV, Guardian (15.6.2021), available at https://www.theguardian.com/world/2021/jun/15/hungary-passes-law-banning-lbgt-content-in-schools. See also Council of Europe Commissioner for Human Rights, Memorandum on the stigmatisation of LGBTI people in Poland, CommDH(2020)27, 3 Dec. 2020, para. 42, available at https://rm.coe.int/memorandum-on-the-stigmatisation-of-lgbti-people-in-poland/1680a08b8e; Romania must reject Hungary-style anti-LGBT bill, rights groups say, Reuters (29.4.2022), available at https://www.reuters.com/world/europe/romania-must-reject-hungary-style-anti-lgbt-bill-rights-groups-say-2022-04-29/.
- 22. See, e.g., Parliamentary Assembly of the Council of Europe (PACE), Empowering women: promoting access to contraception in Europe, Resolution 2331 (2020) and Explanatory Memorandum by Ms Petra Bayr, rapporteur, Doc. 15084, 19 Feb. 2020; European Parliamentary Forum for Sexual and Reproductive Rights, Ensure Sexual and Reproductive Health and Rights Even in Times of Crisis: Report on Access to Contraception in Europe During the COVID-19 Pandemic (2021), available at https://www.epfweb.org/sites/default/files/2021-09/EC\_Report\_v10.pdf; Michielsen, K., Campbell, L., De la Torre, F. F., The impact of COVID-19 on sexual and reproductive health in Eastern Europe and Central Asia, Technical report for UNFPA Regional Office for Eastern Europe and Central Asia (2022), available at https://eeca.unfpa.org/en/publications/impact-covid-19-sexual-and-reproductive-health-eastern-europe-and-central-asia.
- 23. See, e.g., La « pilule du lendemain » est prise en charge à 100 % sans ordonnance, Service-Public.fr (9.2.2023), available at https://www.service-public.fr/particuliers/actualites/A16291; Long-term contraceptives should be cheaper for all women, Belgian MP argues, The Brussels Times (21.2.2023), available at https://www.brusselstimes.com/376369/long-term-contraception-should-be-cheaper-for-all-women-belgian-mp-argues; Free contraception extended to women aged up to 31 in Ireland, The Irish Post (2.1.2024), available at https://www.irishpost.com/news/free-contraception-extended-to-women-aged-up-to-31-in-ireland-265858; European Parliamentary Forum for Sexual and Reproductive Rights, Luxembourg announces 100% reimbursement for contraceptives with no age limit an important step towards Universal Health Coverage!, News (10.5.2023), available at https://www.epfweb.org/node/991.
- 24. See, e.g., PACE, Empowering women: promoting access to contraception in Europe, Resolution 2331 (2020) and Explanatory Memorandum by Ms Petra Bayr, rapporteur, Doc. 15084, 19 Feb. 2020; European Parliamentary Forum for Sexual and Reproductive Rights, Contraception Policy Atlas Europe 2024, available at https://www.epfweb.org/node/1042; Contraceptive Use and Awareness Among Young People in the European Region (2022), available at https://www.epfweb.org/sites/default/files/2022-10/CC%20 AwarenessReport\_final.pdf.

- 25. See, e.g., European Parliamentary Forum for Sexual and Reproductive Rights, Contraception Policy Atlas Europe 2024, https://www.epfweb.org/node/1042; Contraceptive Use and Awareness Among Young People in the European Region (2022), https://www.epfweb.org/sites/default/files/2022-10/CC%20AwarenessReport\_final.pdf; UNFPA Eastern Europe and Central Asia Regional Office (UNFPA EECARO), Getting to zero unmet need for family planning in Eastern Europe and Central Asia (2022), available at https://eeca.unfpa.org/sites/default/files/pub-pdf/fp-rhcs\_thematic\_briefs\_eeca.pdf; European Institute for Gender Equality, Gender Equality Index 2021: Health (Rights, access and outcomes sexual and reproductive health in focus), available at https://eige.europa.eu/publications-resources/toolkits-guides/gender-equality-index-2021-report/rights-access-and-outcomes-sexual-and-reproductive-health-focus?language\_content\_entity=en.
- 26. European Consortium for Emergency Contraception, *An update on access to emergency contraception in Europe*, Feb. 2022, available at https://www.ec-ec.org/wp-content/uploads/2022/02/ECEC-Update-EC-access-Europe\_FEB\_2022\_FINAL.pdf.
- 27. Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual and reproductive health and rights in Europe* (2017), p. 58, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead.
- See, e.g., CESCR, General comment No. 25 (2020) on science and economic, social and cultural rights (article 15 (1) (b), (2), (3) and (4) of the International Covenant on Economic, Social and Cultural Rights), E/C.12/GC/25 (2020), para. 33; CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), para. 42(a); Concluding Observations: Latvia, E/C.12/LVA/CO/2 (2021), para. 42-43; CEDAW, Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(b); Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), para. 37(b); Concluding Observations: Georgia, CEDAW/C/GEO/CO/6 (2023), para. 38(b); Concluding Observations: Switzerland, CEDAW/C/CHE/CO/6 (2022), para. 56(b); CRC, Concluding Observations: Poland, CRC/C/POL/CO/5-6 (2021), para. 36(c); Concluding Observations: Malta, CRC/C/MLT/ CO/3-6 (2019), para. 33(b); Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to Austria from 13 to 17 December 2021, para. 103, CommDH(2022)10, available at https://search.coe.int/commissioner/Pages/ result details.aspx?ObjectId=0900001680a6679a. See also PACE, Empowering women: promoting access to contraception in Europe, Resolution 2331 (2020) and Explanatory Memorandum by Ms Petra Bayr, rapporteur, Doc. 15084, 19 Feb. 2020.
- 29. CRC, Concluding Observations: Ireland, CRC/C/IRL/CO/5-6 (2023), para. 33(c); Concluding Observations: Poland, CRC/C/POL/CO/5-6 (2021), para. 36(a)(c); Committee on the Rights of Persons with Disabilities (CRPD), Concluding Observations: Spain, CRPD/C/ESP/CO/2-3, para. 49.
- 30. CEDAW, Concluding Observations: Switzerland, CEDAW/C/CHE/CO/6 (2022), para. 56(b).
- 31. CRPD, Concluding Observations: Poland, CRPD/C/POL/CO/1 (2018), paras. 10(c), 44(e); Concluding Observations: Hungary, CRPD/C/HUN/CO/2-3 (2022), para. 36; Concluding Observations: Georgia, CRPD/C/GEO/CO/1 (2023), paras. 26, 36.
- 32. Cyprus: ΟπερίΠοινικού Κώδικα Νόμος (ΚΕΦ.154) N.23(I)/2018 NOMOΣΠΟΥΤΡΟΠΟΠΟΙΕΙ ΤΟΝ ΠΟΙΝΙΚΟ ΚΩΔΙΚΑ (Penal Code, Chapter 154 as modified by Law 23(I)/2018 amending the penal code), available at http://cylaw.org/nomoi/enop/ind/0\_154/section-sc4a07be8d-721f-4bc3-9eae-eddd79e0885e-In33e6cfc1-3ae9-4878-9a51-0ca801e8bc27.html; Finland: Lag om avbrytande av havandeskap 239/1970 (Act on the Termination of Pregnancy n° 239/1970 as amended), available at https://finlex.fi/sv/laki/ajantasa/1970/19700239?search[type]=pika&search[pika]=abort#a20.12.2022-1097;

- Iceland: Lög um þungunarrof, 2019 nr. 43 22. Maí (Abortion law 2019 No. 43 May 22), available at https://www.althingi.is/altext/149/s/0521.html; Ireland: Health (Regulation of Termination of Pregnancy) Act 2018, available at https://data.oireachtas.ie/ie/oireachtas/act/2018/31/eng/enacted/a3118.pdf; Northern Ireland: Abortion (Northern Ireland) Regulations 2020, available at https://www.legislation.gov.uk/uksi/2020/345/made; San Marino: Legge 7 settembre 2022 n.127 Regolamentazione dell'interruzione volontaria di gravidanza (Law no. 7 of 2022 September 127 Regulation of the voluntary interruption of pregnancy), available at https://www.consigliograndeegenerale.sm/on-line/home/archivio-leggi-decreti-e-regolamenti/scheda17177841.html.
- Belgium: Loi relative à l'interruption volontaire de grossesse, abrogeant les articles 350 et 351 du Code pénal et modifiant les articles 352 et 383 du même Code et modifiant diverses dispositions législative (Law on the voluntary termination of pregnancy, abrogating articles 350 and 351 of the Penal code and modifying articles 352 and 383 of the same Code and modifying various legislative provisions), available at https://www. ejustice.just.fgov.be/cgi\_loi/change\_lg.pl?language=fr&la=F&cn=2018101503&table\_ name=loi; France: LOI n° 2022-295 du 2 mars 2022 visant à renforcer le droit à l'avortement (Law N° 2022-295 of 2 March 2022 seeking to strengthen the right to abortion), available at https://www.vie-publique.fr/loi/276586-loi-2-mars-2022renforcer-droit-avortement-delai-porte-14-semaines; Lithuania: JSAKYMASDĖL NĖŠTUMO NUTRAUKIMO PASLAUGŲ ORGANIZAVIMO IR TEIKIMO TVARKOS APRAŠO PATVIRTINIMO 2022 m. rugpjūčio 12 d. Nr. 1348 (Order on the appoval of the description of the procedure for the organisation and provision of termination of pregnancy services 12 Aug. 2022 No. 1348), available at https://www.e-tar.lt/portal/ lt/legalAct/5cad1e201a1811edb4cae1b158f98ea5; Moldova: Ordinul Ministerului Sănătății, Muncii și Protecției Sociale al Republicii Moldova nr.766 din 18.08.2020 Cu privire la aprobarea Standardului privind efectuarea întreruperii sarcinii în conditii de siguranță (Order of the Ministry of Health, Labour and Social Protection of the Republic of Moldova no.766 of 18.08.2020 On the approval of the Standard on Safe Termination of Pregnancy), available at http://www.cidsr.md/wp-content/uploads/2020/09/ Standardul-privind-%C3%AEntreruperea-sarcinii-%C3%AEn-condi%C8%9Bii-desiguran%C8%9B%C4%83-aprobat-prin-ordinul-MSMPS-nr.766-din-18.08.2020.pdf; Netherlands: Wet van 22 augustus 2022, houdende voorstel van wet van de leden Paternotte, Kuiken, Ellemeet en Tellegen tot wijziging van Wet afbreking zwangerschap in verband met het afschaffen van de verplichte minimale beraadtermijn voor de afbreking van zwangerschappen (Act of August 22, 2022, amending the Termination of Pregnancy Act in connection with the abolition of the compulsory minimum consideration period for the termination of pregnancies), available at https://zoek. officielebekendmakingen.nl/stb-2022-326.html; North Macedonia: 3AKOH ПРЕКИНУВАЊЕ НА БРЕМЕНОСТА бр. 101/2019 од 22.5.2019 година (Law of termination of pregnancy no. 101/2019 of 22.5.2019), available at https://hera.org.mk/wp-content/ uploads/2019/12/ZAKON-ZA-PREKINUVANE-NA-BREMENOST-2019.pdf; Spain: Ley Orgánica 4/2022, de 12 de abril, por la que se modifica la Ley Orgánica 10/1995, de 23 de noviembre, del Código Penal, para penalizar el acoso a las mujeres que acuden a clínicas para la interrupción voluntaria del embarazo (Organic Law 4/2022, of April 12, which amends Organic Law 10/1995, of November 23, 1995, of the Penal Code, to criminalize the harassment of women who go to clinics for the voluntary termination of pregnancy), available at https://www.boe.es/eli/es/lo/2022/04/12/4 and Ley Orgánica 1/2023, de 28 de febrero, por la que se modifica la Ley Orgánica 2/2010, de 3 de marzo, de salud sexual y reproductiva y de la interrupción voluntaria del embarazo (Organic Law 1/2023, of February 28, which amends Organic Law 2/2010, of March 3, on sexual and reproductive health and the voluntary interruption of pregnancy), available at https://www.boe.es/diario\_boe/txt.php?id=BOE-A-2023-5364; England, Scotland and

- Wales: Department of Health and Social Care and Maggie Throup MP, At home early medical abortions made permanent in England and Wales (23 August 2022), available at <a href="https://www.gov.uk/government/news/at-home-early-medical-abortions-made-permanent-in-england-and-wales">https://www.gov.uk/government/news/at-home-early-medical-abortions-made-permanent-in-england-and-wales</a> and Letter from the Chief Medical Officer For Scotland Professor Sir Gregor Smith: Abortion Additional approval for Mifepristone to be taken at home (11 Sept. 2023), available at <a href="https://www.sehd.scot.nhs.uk/cmo/CMO(2023)16">https://www.sehd.scot.nhs.uk/cmo/CMO(2023)16</a>. pdf.
- 34. Belgium: Belgian Green MPs table abortion reform proposal, Euractiv (29.9.2023), available at https://www.euractiv.com/section/politics/news/belgian-green-mpstable-abortion-reform-proposal/; Denmark: Danish ethics council recommends changing limit on abortion, The Local (26.9.2023), available at https://www.thelocal. dk/20230926/danish-ethics-council-recommends-changing-limit-on-abortion; France: L'Assemblée nationale vote en première lecture l'inscription de l'IVG dans la Constitution, Le Monde (30.1.2024), available at https://www.lemonde.fr/societe/ article/2024/01/30/l-assemblee-nationale-vote-en-premiere-lecture-l-inscription-del-vg-dans-la-constitution\_6213916\_3224.html; Germany: Kommission zur reproduktiven Selbstbestimmung und Fortpflanzungsmedizin konstituiert sich, Bundesministerium der Justiz, Press Release (31.3.2023), available at https://www.bmj.de/SharedDocs/ Pressemitteilungen/DE/2023/0331\_Kommission\_Selbstbestimmung.html; Oireachtas, Health Committee supports findings of Termination of Pregnancy Review, Houses of the Oireachtas (15.12.2023), available at https://www.oireachtas.ie/en/ press-centre/press-releases/20231215-health-committee-supports-findings-oftermination-of-pregnancy-review/; the Netherlands: GroenLinks-PvdA submit bill to remove abortion from Criminal Code, NL Times (15.11.2023), available at https:// nltimes.nl/2023/11/15/groenlinks-pvda-submit-bill-remove-abortion-criminalcode; Norway: Norwegian government to review abortion laws, The Local (1.6.2022), available at https://www.thelocal.no/index.php/20220601/norwegian-governmentto-review-abortion-laws; Poland: Half of public support Tusk's aim to introduce abortion on demand in Poland, find polls, Notes from Poland (2.2.2024), available at https:// notesfrompoland.com/2024/02/02/half-of-public-support-tusks-aim-to-introduceabortion-on-demand-in-poland-find-polls/; the United Kingdom: MPs Could Vote On Decriminalising Abortion In England And Wales, PoliticsHome (27.11.2023), available at https://www.politicshome.com/news/article/mps-vote-decriminalising-abortionengland-wales-stella-creasy.
- 35. See, e.g., Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual and reproductive health and rights in Europe* (2017), pp. 47-56, 59, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead; CEDAW, *Inquiry concerning the United Kingdom of Great Britain and Northern Ireland under article 8 of the Optional Protocol to the Convention on the Elimination of All Forms of Discrimination against Women*, CEDAW/C/OP.8/GBR/1 (2018); Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to Malta from 11 to 16 October 2021, CommDH(2022)1, paras. 96-97, available at https://rm.coe.int/report-of-the-council-of-europe-commissioner-for-human-rights-dunja-mi/1680a5498d.
- 36. CEDAW, Concluding Observations: Guatemala, CEDAW/C/GTM/CO/10 (2023), para. 39(a).
- 37. Human Rights Committee (HRC), General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8.
- 38. See, e.g., CEDAW, Concluding Observations: Guatemala, CEDAW/C/GTM/CO/10 (2023), para. 39(a); Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(d); CRC, Concluding Observations: Malta, CRC/C/MLT/CO/3-6 (2019), para. 33(c);

- Concluding Observations: Poland, CRC/C/POL/CO/5-6 (2021), para. 36(d); HRC, General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8; Concluding Observations: Ireland, CCPR/C/IRL/CO/5 (2023), para. 26(a).
- 39. CEDAW, Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(d); Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), para. 37(c); Concluding Observations: Monaco, CEDAW/C/MCO/CO1-3 (2017), para. 42(c); CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), para. 42(b)(d); Spain, E/C.12/ESP/CO/6 (2018), para. 44(d).
- 40. HRC, Concluding Observations: Ireland, CCPR/C/IRL/CO/5 (2023), para. 26(d); CEDAW, Concluding Observations: Spain, CEDAW/C/ESP/CO/9 (2023), para. 35(b); CRC, Concluding Observations: Croatia, CRC/C/HRV/CO/5-6 (2022), para. 35(b); Concluding Observations: North Macedonia, CRC/C/MKD/CO/3-6 (2022), para. 33(d).
- 41. See, e.g., CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), para. 42(a); Concluding Observations: Republic of Moldova, E/C.12/MDA/CO/3 (2017), para. 63(c); CEDAW, Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(d); Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), para. 37(b); Concluding Observations: Belgium, CEDAW/C/BEL/CO/8 (2022), para. 46(a); Concluding Observations: Republic of Moldova, CEDAW/C/MDA/CO/6 (2020), para. 39(c); Concluding Observations: Austria, CEDAW/C/AUT/CO/9 (2019), para. 35(b).
- 42. CEDAW, Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(c).
- 43. CESCR, General comment No. 25 (2020) on science and economic, social and cultural rights (article 15 (1) (b), (2), (3) and (4) of the International Covenant on Economic, Social and Cultural Rights), E/C.12/GC/25 (2020), para. 33; Women's and girls' sexual and reproductive health rights in crisis, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), para. 77(b); CEDAW, Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(c); , Concluding Observations: Austria, CEDAW/C/AUT/CO/9 (2019), para. 35(b). Human rights mechanisms have welcomed measures to expand access to self-managed medication abortion, including through telemedicine, during the COVID-19 pandemic and highlighted that states are obliged to quarantee access to abortion care during times of crisis. See Women's and girls' sexual and reproductive health rights in crisis, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), paras. 20, 29; Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, Tlaleng Mofokeng, Sexual and reproductive health rights: challenges and opportunities during the COVID-19 pandemic, A/76/172 (2021), paras. 23, 40-43.
- 44. See, e.g., HRC, General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8; CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), paras. 41, 42(e).
- 45. See, e.g., HRC, General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8; Concluding Observations: Romania, CCPR/C/ROU/CO/5 (2017), para. 26; Concluding Observations: Ireland, CCPR/C/IRL/CO/5 (2023), para. 26(b); CECSR, Concluding Observations: Italy, E/C.12/ITA/CO/6 (2022), paras. 57-58; Concluding Observations: Spain, E/C.12/ESP/CO/6 (2018), paras. 43, 44(b); CEDAW, Concluding Observations: Austria, CEDAW/C/AUT/CO/9 (2019), paras. 34(c), 35(c).
- 46. European Court of Human Rights, *Grimmark v. Sweden*, Application No. 43726/17, decision of 11 February 2020, paras. 25-26; *Steen v. Sweden*, Application No. 62309/17, decision of 11 February 2020, paras. 20-21.
- 47. HRC, Concluding Observations: Ireland, CCPR/C/IRL/CO/5 (2023), para. 26(e). See also Human Rights Committee (HRC), General comment No. 36 Article 6: right to life,

- CCPR/C/GC/36 (2019), para. 8; PACE, Access to abortion in Europe: stopping anti-choice harassment, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022.
- 48. Report of the Special Rapporteur of the Human Rights Council on extrajudicial, summary or arbitrary executions, Agnes Callamard, *Saving lives is not a crime*, A/73/314 (2018), para. 89(h).
- 49. World Health Organization (WHO), Abortion care guideline (2022). See also WHO, Clinical practice handbook for quality abortion care (2023).
- 50. Ibid., p. xx.
- 51. Ibid., pp. xx, 21-29.
- 52. Ibid., p. xxi.
- 53. Ibid., pp. 24-25.
- 54. Ibid., pp. 26-27, 42-44.
- 55. Ibid., pp. 26-29, 41-42.
- 56. Ibid., pp. 95-100.
- 57. Ibid., p. 59.
- 58. Ibid., pp. 60-61.
- 59. Andorra does not allow abortion at all. Recent legislation in Malta provides for access to abortion only when the life of a pregnant woman is at risk. Liechtenstein and Poland allow abortion only when a pregnant woman's life or health is at risk, or the pregnancy is the result of sexual violence. In Monaco, abortions are legal only when a pregnant woman's life or health is at risk, the pregnancy is the result of sexual violence or involves a severe fetal impairment. See Andorra: Llei 9/2005, del 21 de febrer, qualificada del Codi penal, Arts. 107-109 (Law 9/2005, 21 February, acting as Penal code, Arts. 107-109); Malta: KODIĊI KRIMINALI, KAPITOLU 9, 243B (Criminal Code, Chapter 9, Art. 243B); Liechtenstein: Strafgesetzbuch vom 24. Juni 1987 (StGB), Abschnitt 2 (Criminal Code of 24 June 1987 (StGB), Sec. 2); Poland: USTAWA z dnia 7 stycznia 1993 r. o planowaniu rodziny, ochronie płodu ludzkiego i warunkach dopuszczalności przerywania ciąży (Act on Family Planning, Protection of the Human Fetus, and Conditions for Pregnancy Termination of 7 Jan. 1993); Monaco: Code pénal, Article 248 (Penal code, Article 248).
- 60. CEDAW, Inquiry concerning the United Kingdom of Great Britain and Northern Ireland under article 8 of the Optional Protocol to the Convention on the Elimination of All Forms of Discrimination against Women, CEDAW/C/OP.8/GBR/1 (2018), paras. 22, 64; WHO, Abortion care guideline (2022), page 26-27.
- 61. Belgium: Loi relative à l'interruption volontaire de grossesse, abrogeant les articles 350 et 351 du Code pénal et modifiant les articles 352 et 383 du même Code et modifiant diverses dispositions législatives, 15 octobre 2018, Article 3 (Law on the voluntary termination of pregnancy, repealing articles 350 and 351 of the penal code and amending articles 352 and 383 of the same code and amending various legislative provisions, 15 October 2018, Article 3), available at http://www.ejustice.just.fgov.be/cgi\_loi/change\_lg.pl?language=fr&la=F&cn=2018101503&table\_name=loi; Cyprus: Ο περί Ποινικού Κώδικα Νόμος (ΚΕΦ.154) Ν. 23(I)/2018 ΝΟΜΟΣ ΠΟΥ ΤΡΟΠΟΠΟΙΕΙ ΤΟΝ ΠΟΙΝΙΚΟ ΚΩΔΙΚΑ, 168 (Penal Code, Chapter 154 as modified by Law 23(I)/2018 amending the penal code, Article 168), available at http://cylaw.org/nomoi/enop/ind/0\_154/section-scf87db566-babb-4f92-9d6b-b3d3303615c3.html; Iceland: Almenn hegningarlög, 216. gr. (General Criminal law, Article 216), available at https://www.althingi.is/lagas/nuna/1940019.html; Italy: Legge 22 maggio 1978, n. 194, Norme per la tutela sociale della maternita' e sull'interruzione volontaria della gravidanza, Gazzetta Ufficiale della Repubblica Italiana No. 140 (22 maggio 1978) (Law No. 194 of May 22,

1978, Provisions on the Social Protection of Maternity and the Voluntary Interruption of Pregnancy, Official Gazette No. 140 (May 22, 1978), available at https://www.antonellapedone.com/leggi/legge-194-1978-05-22#:~:text=Legge%20n.194%20 del%2022%20maggio%201978%20Norme%20per,e%20tutela%20la%20vita%20 umana%20dal%20suo%20inizio; Liechtenstein: Article 96 para 3 of the Criminal Code of Liechtenstein, 24 June 1987, available at https://www.gesetze.li/konso/html/1988037000?version=53&search\_text=&search\_loc=#par:96; Portugal: Código Penal, Decreto-Lei No. 48/1995, versão consolidada (Penal Code, Decree-Law no. 48/95 as amended), Articles 140-142, available at https://diariodarepublica.pt/dr/legislacao-consolidada/decreto-lei/1995-34437675; San Marino: Legge 7 settembre 2022 n.127 - Regolamentazione dell'interruzione volontaria di gravidanza, articolo 16 che modifica l'articolo 153 del Codice penale (Law no. 7 of 2022 September 127 - Regulation of the voluntary interruption of pregnancy, Article 16 amending Article 153 of the Criminal Code), available at https://www.consigliograndeegenerale.sm/on-line/home/archivio-leggi-decreti-e-regolamenti/scheda17177841.html.

- 62. See, e.g., WHO, Abortion care guideline (2022), pp. 22, 24-25.
- 63. CEDAW, Inquiry concerning the United Kingdom of Great Britain and Northern Ireland under article 8 of the Optional Protocol to the Convention on the Elimination of All Forms of Discrimination against Women, CEDAW/C/OP.8/GBR/1 (2018), paras. 30, 33-42, 83.
- 64. See, e.g., FEDERA, SRHR and FEDERA Overview June-July 2023, available at https://en.federa.org.pl/srhr-and-federa-overview-june-july-2023/; Komentarz FEDERY do postawienia zarzutów dr Marii Kubisie, 27.11.2023, available at https://federa.org.pl/wp-content/uploads/2023/11/DrKubisa\_stanowisko\_FEDERY.pdf; Communication from NGOs (Foundation for Women and Family Planning and the Center for Reproductive Rights) (18/04/2023) in the cases of R.R., TYSIAC and P. and S. v. Poland (Applications No. 27617/04, 5410/03, 57375/08), 1468th meeting (June 2023) (DH), available at https://hudoc.exec.coe.int/?i=DH-DD(2023)525E.
- 65. See Center for Reproductive Rights, *European Abortion Laws: A Comparative Overview* (Sept. 2023), available at https://reproductiverights.org/european-abortion-laws-comparative-overview/.
- 66. WHO, Global Abortion Policies Database, available at https://abortion-policies.srhr.org/countries/?r%5B%5D=r13&f%5B2%5D%5B%5D=8&hrt=&co=&pia=; ASTRA Network, The Fight Hidden in Plain Sight: Sexual and Reproductive Health and Rights in Central and Eastern Europe and Central Asia (2020), page 45, 92, https://astra.org.pl/wp-content/uploads/2020/02/The-fight-hidden-in-plain-sight.pdf?fbclid=lwAR3JS3Wbh2ns-oLggN5kpLOmo5\_yDUnECCHo-ntgj8XfuM8h9EMaU8EY6Dk.
- 67. See, e.g., WHO, Abortion care guideline (2022), pp. 12-13, 28-29, 41-44.
- 68. Lack of access to abortion services guaranteed by law: the case of Mirela Čavajda in Croatia: A submission by the DOSTA! ENOUGH! Initiative Croatia (2022), available at https://www.roda.hr/media/attachments/english\_roda/Reports/Final\_Communication\_to\_SP.pdf; Council of Europe Commissioner for Human Rights, Report following her visit to Italy from 19 to 23 June 2023, CommDH(2023)37, paras. 122-124, available at https://rm.coe.int/country-visit-report-on-italy-following-the-visit-from-19-to-23-june-2/1680adae5a; Women's rights denied: Abortion on the line as Italy's far right eyes power, France 24 (23.9.2022), available at https://www.france24.com/en/europe/20220923-women-s-rights-denied-abortion-on-the-line-as-italy-s-far-right-eyes-power; Amnesty International, Legal but not local: Barriers to accessing abortion services in Northern Ireland (2023), pp. 81-92, available at https://www.amnesty.org.uk/barriers-abortion-access-northern-ireland-0; European Parliament resolution of 11 November 2021 on the first anniversary of the de facto abortion ban in Poland (2021/2925(RSP)), para. 15, available at https://www.europarl.europa.eu/doceo/document/TA-9-2021-

0455\_EN.html; FILIA Center, Refusal to Perform Abortion on Request in Romania (2020-2021), available at https://centrulfilia.ro/new/wp-content/uploads/2021/10/Abortion-report.pdf; Holubová B. et al., Dostupnosť služieb reprodukčného zdravia na Slovensku - Správa o poskytovateľoch zdravotnej starostlivosti [Reproductive healthcare availability in Slovakia - Report on healthcare providers], Možnosť voľby (2021), pp. 66-69, available at http://moznostvolby.sk/dostupnost-sluzieb-reprodukcneho-zdravia-na-slovensku-2/.

- 69. See, e.g., WHO, Abortion care guideline (2022), p. 61.
- 70. See, e.g., Amnesty International, *Legal but not local: Barriers to accessing abortion services in Northern Ireland* (2023), available at https://www.amnesty.org.uk/barriers-abortion-access-northern-ireland-0; PACE, *Access to abortion in Europe: stopping anti-choice harassment*, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022; Zanini G. et al., *Abortion information governance and women's travels across European borders*, Women's Studies International Forum, 87 (2021) 102496.
- 71. European Parliamentary Forum for Sexual and Reproductive Rights, International Planned Parenthood Federation European Network (IPPF EN), *European Abortion Policies Atlas 2021*, available at https://www.epfweb.org/sites/default/files/2022-10/ABORT%20Atlas EN%202021-v10%20%281%29.pdf.
- 72. For example: Albania, Cyprus, Greece, and Hungary. See Albania: LIGJI Nr.8045, dt 7.12.1995 PËR NDËRPRERJEN E SHTATËZANISË Kreu III Neni 15 (Law number 8045 dated December 7, 1995, On the Interruption of Pregnancy, Chapter III article 15); Cyprus: Ο περί Ποινικού Κώδικα Νόμος (ΚΕΦ.154) N. 23(I)/2018 NOMOΣ ΠΟΥ ΤΡΟΠΟΠΟΙΕΙ ΤΟΝ ΠΟΙΝΙΚΟ ΚΩΔΙΚΑ, 168B (Penal Code, Chapter 154 as modified by Law 23(I)/2018 amending the penal code, Article 169B); Greece: Poinikos Kodikas 305 (Criminal Code 305); Hungary: 2008. évi XLVIII. Törvény a gazdasági reklámtevékenység alapvető feltételeiről és egyes korlátairól 17 (Act XLVIII of 2008 on the Basic Requirements and Certain Restrictions of Commercial Advertising Activities 17). See also 1992. évi LXXIX. törvény a magzati élet védelméről 15 (Act LXXIX of 1992 Law for the Protection of Fetal Life 15).
- 73. See, e.g., PACE, *Access to abortion in Europe: stopping anti-choice harassment*, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022, para. 19.
- 74. Slovakia joint civil society letter (22 Oct. 2021), available at https://reproductiverights. org/slovakia-joint-civil-society-letter/.
- 75. Hoctor L., Lamačková A., Thomasen K., *Anti-Equality and Anti-SRHR Backlash in the EU*, in: Centre for Feminist Foreign Policy, *Power over Rights: Understanding and countering the transnational anti-gender movement, Volume II: Case Studies* (March 2021), pp. 55-56, available at <a href="https://centreforffp.net/wordpress/wp-content/uploads/2023/01/PowerOverRights2\_web.pdf">https://centreforffp.net/wordpress/wp-content/uploads/2023/01/PowerOverRights2\_web.pdf</a>.
- 76. Constitutional Tribunal of Poland, Case no. K 1/20 (2020).
- 77. See Notification of 12 applications concerning abortion rights in Poland, Press Release, ECHR 217 (2021), available at https://hudoc.echr.coe.int/eng-press?i=003-7074470-9562874.
- 78. See, e.g., *Sprawa Izabeli z Pszczyny. Śledztwo zostało przedłużone*, Puls Medycyny (18.5.2023) available at https://pulsmedycyny.pl/sprawa-izabeli-z-pszczyny-sledztwo-zostalo-przedluzone-1185666; *Mąż zmarłej Ani ze Świdnicy: Nie rozumiem, dlaczego kazali jej rodzić martwe dziecko*, Wyborcza.pl (13.11.2021) available at https://walbrzych. wyborcza.pl/walbrzych/7,178336,27790843,ania-ze-swidnicy-wywiad.html; *37-letnia Agnieszka z Częstochowy zmarła po stracie bliźniaczej ciąży. Jest stanowisko RPP*, poradnikzdrowie.pl (28.7.2023), available at https://www.poradnikzdrowie.pl/aktualnosci/37-letnia-agnieszka-z-czestochowy-zmarla-po-stracie-blizniaczej-ciazy-

- jest-stanowisko-rpp-aa-4Snp-cRJE-tXfX.html; Polish authorities rule rights of pregnant woman who died were violated amid abortion law debate, Associated Press (12.6.2023), available at https://apnews.com/article/poland-abortion-pregnant-woman-died-bc4a27a9cd8df2728e01bf77c995b70f.
- 79. European Court of Human Rights, *M.L. v. Poland*, Application No. 40119/21, judgment of 14 December 2023.
- 80. Hungary tightens abortion access with listen to 'foetal heartbeat' rule, The Guardian (13.9.2022) available at https://www.theguardian.com/global-development/2022/sep/13/hungary-tightens-abortion-access-with-listen-to-foetal-heartbeat-rule.
- 81. "Rights defenders raise concerns as Georgia introduces mandatory pre-abortion consultations," OC Media (9.11.2023), available at https://oc-media.org/rights-defenders-raise-concerns-as-georgia-introduces-mandatory-pre-abortion-consultations/; ANSER, Rethinking Abortion Policy in Georgia: From Restrictive Measures to Evidence-Based Approaches (2023), available at https://www.ugent.be/anser/en/anser-position-paper-rethinking-abortion-policy-in-georgia.pdf.
- Italian senator renews anti-abortion foetus rights proposal, The Guardian (19.10.2022), available at https://www.theguardian.com/world/2022/oct/19/italian-senator-renewsanti-abortion-foetus-rights-proposal; "Women mulling abortion may soon be asked to reconsider – new law makes way in Lithuanian parliament," LRT.lt (6.9.2022), available https://www.lrt.lt/en/news-in-english/19/1715331/women-mulling-abortionmay-soon-be-asked-to-reconsider-new-law-makes-way-in-lithuanian-parliament; "'Christian Values' Before Human Rights: Lithuania Debates Abortion Ban," Liberties (29.3.2018), available at https://www.liberties.eu/en/stories/abortion-ban-lithuaniahuman-rights-committee-parliament/14543; Jointsubmission by the Freedom of Choice (Možnosť voľby), InTYMYta and Center for Reproductive Rights to the Committee on the Elimination of Discrimination Against Women's 85th session: Periodic review of Slovakia (April 2023), available at https://tbinternet.ohchr.org/ layouts/15/treatybodyexternal/ Download.aspx?symbolno=INT%2FCEDAW%2FCSS%2FSVK%2F52443&Lang=en; Council of Europe Commissioner for Human Rights, Slovak Republic: Parliament should reject bill that restricts access to safe and legal abortion services (2021), available at https://www.coe.int/en/web/commissioner/-/slovak-republic-parliament-shouldreject-bill-that-restricts-access-to-safe-and-legal-abortion-services.
- 83. Council of Europe Commissioner for Human Rights, Issue Paper on *Women's sexual* and reproductive health and rights in Europe, Issue paper published by the (2017), p. 50, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead.
- 84. CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), paras. 41, 42(e); Human Rights Committee (HRC), General comment No. 36 Article 6: right to life, CCPR/C/GC/36 (2019), para. 8.
- 85. Drandić D. et al., Parent organizations' experiences of the pandemic response in maternity care in thirteen European countries, Eur. J. Midwifery (2022) 22:6:71.
- 86. See, e.g., Report of the Special Rapporteur on violence against women, its causes and consequences, Dubravka Šimonović, A human rights-based approach to mistreatment and violence against women in reproductive health services with a focus on childbirth and obstetric violence, A/74/137 (2019); CEDAW, S.F.M v. Spain, CEDAW/C/75/D/138/2018 (2020); N.A.E. v. Spain, CEDAW/C/82/D/149/2019 (2022); M.D.C.P v. Spain, CEDAW/C/84/D/154/2020 (2023); Concluding Observations: Germany, CEDAW/C/DEU/CO/9 (2023), para. 46(e); Concluding Observations: Bulgaria, CEDAW/C/BGR/CO/8 (2020), para. 34(a)(d). See also Zampas Ch. et al. Operationalizing a Human Rights-Based Approach to Address Mistreatment against Women during Childbirth, Health Hum. Rights

- (2020) 22(1): pp. 251-264; PACE, Obstetrical and gynaecological violence, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019; Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to Spain from 21 to 25 November 2022, CommHR(2023)9, paras. 62-63, 74, available at https://rm.coe.int/report-on-spain-following-the-visit-from-21-to-25-november-2022-by-dun/1680aaeb17.
- 87. Report of the Special Rapporteur on violence against women, its causes and consequences, Dubravka Šimonović, A human rights-based approach to mistreatment and violence against women in reproductive health services with a focus on childbirth and obstetric violence, A/74/137 (2019), paras. 4, 9; see also CEDAW, N.A.E. v. Spain, CEDAW/C/82/D/149/2019 (2022), para. 15.4; M.D.C.P. v. Spain, CEDAW/C/84/D/154/2020 (2023), para. 7.8.
- 88. Report of the Special Rapporteur on violence against women, its causes and consequences, Dubravka Šimonović, A human rights-based approach to mistreatment and violence against women in reproductive health services with a focus on childbirth and obstetric violence, A/74/137 (2019), paras. 8, 11, 31.
- 89. Ibid., paras. 39-49.
- 90. Ibid., paras. 75-76.
- 91. *S.F.M. v. Spain*, CEDAW/C/75/D/138/2018 (2020); *N.A.E. v. Spain*, CEDAW/C/82/D/149/2019 (2022); *M.D.C.P. v. Spain*, CEDAW/C/84/D/154/2020 (2023).
- 92. *N.A.E. v. Spain*, CEDAW/C/82/D/149/2019 (2022), para. 15.3; *M.D.C.P. v. Spain*, CEDAW/C/84/D/154/2020 (2023), para. 7.7.
- 93. *N.A.E.* v. *Spain*, CEDAW/C/82/D/149/2019 (2022), paras. 15.4, 16(b)(iii); *M.D.C.P.* v. *Spain*, CEDAW/C/84/D/154/2020 (2023), para. 7.8; *S.F.M.* v. *Spain*, CEDAW/C/75/D/138/2018 (2020), para. 8(b)(iii).
- 94. See, e.g., CEDAW, Concluding Observations: Iceland, CEDAW/C/ISL/CO/9 (2023), paras. 23-24; Concluding Observations: Belgium, CEDAW/C/BEL/CO/8 (2022), paras. 24(b), 42, 46; CRPD, Concluding Observations: Hungary, CRPD/C/HUN/CO/2-3 (2022), paras. 35-36; Concluding Observations: Norway, CRPD/C/NOR/CO/1 (2019), para. 30; Committee on the Elimination of Racial Discrimination (CERD), Concluding Observations: Slovakia, CERD/C/SVK/CO/13 (2022), para. 31.
- 95. European Court of Human Rights, *G.M. and Others v. the Republic of Moldova*, Application No. 44394/15, judgment of 22 November 2022; *S.F.K. v. Russia*, Application No. 5578/12, judgment of 11 October 2022.
- 96. WHO, WHO recommendations: intrapartum care for a positive childbirth experience (2018), available at https://www.who.int/publications/i/item/9789241550215.
- 97. See, e.g., Baranowska B. et al., *Is there respectful maternity care in Poland? Women's views about care during labor and birth*, BMC Pregnancy and Childbirth (2019) 19:520; RODA Parents in Action, *Report to the United Nations Special Rapporteur on Violence Against Women in response to her Call for Submissions, due 17 May 2019*, available at https://www.ohchr.org/sites/default/files/Documents/Issues/Women/SR/ReproductiveHealthCare/Roda\_-\_Parents\_in\_Action\_Croatia.pdf; Public Defender of Rights of the Slovak Republic, Mária Patakyová, *Hovorme otvorene o pôrodoch: ľudskoprávny prístup pri poskytovaní zdravotnej starostlivosti pri pôrodoch (Let's talk openly about childbirth: a human rights approach to the provision of health care during childbirth) (2021), available at https://vop.gov.sk/wp-content/uploads/2021/09/Sprava\_porody\_FINAL.pdf; Mena-Tudela D. et al., <i>Obstetric Violence in Spain (Part I): Women's Perception and Interterritorial Differences*, Int. J. Environ. Res. Public Health (2020) 17:7726; Annborn A., Finnbogadóttir H.R., *Obstetric violence a qualitative interview study*, Midwifery (2022)

- 105:103212; PACE, *Obstetrical and gynaecological violence*, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019.
- 98. See, e.g., PACE, Obstetrical and gynaecological violence, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019; IPPF European Network, Gynaecological and obstetric violence (2022), available at https://europe.ippf.org/sites/europe/files/2022-12/Gynaecological%20 and%20Obstetric%20Violence%20-%20IPPF%20EN%20Research%20%26%20 Policy%20Paper.pdf; ASTRA Network, The Fight Hidden in Plain Sight: Sexual and Reproductive Health and Rights in Central and Eastern Europe and Central Asia (2020), p. 48, available at https://astra.org.pl/wp-content/uploads/2020/02/The-fight-hidden-in-plain-sight.pdf?fbclid=lwAR3JS3Wbh2ns-oLggN5kpLOmo5\_yDUnECCHontgj8XfuM8h9EMaU8EY6Dk;HasanbegovićVukas D., Research on Regulation, Availability and Practice of Abortion in Bosnia and Herzegovina (2023), available at https://soc.ba/site/wp-content/uploads/2023/10/ENG-lstrazivanje-o-regulaciji-dostupnosti-i-praksi-abortusa-u-BIH-WEB-2023\_9\_21.pdf; Haut Conseil à l'Egalité entre les femmes et les hommes, Actes sexistes durant le suivi gynécologique et obstétrical : reconnaître et mettre fin à des violences longtemps ignorées, 29 June 2018.
- 99. See, e.g., Report of the Special Rapporteur on violence against women, its causes and consequences, Dubravka Šimonović, *A human rights-based approach to mistreatment and violence against women in reproductive health services with a focus on childbirth and obstetric violence*, A/74/137 (2019), paras. 21, 28, 31, 44, 45; PACE, *Obstetrical and gynaecological violence*, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019.
- 100. See, e.g., PACE, Obstetrical and gynaecological violence, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019, paras. 23-24; S.F.M v. Spain, CEDAW/C/75/D/138/2018 (2020), paras. 2.8, 2.9; N.A.E. v. Spain, CEDAW/C/82/D/149/2019 (2022), paras. 2.15-2.17, 15.6; M.D.C.P v. Spain, CEDAW/C/84/D/154/2020 (2023), paras. 2.6, 3.6, 7.6, 7.10; Henriksen L. et al., Adaption and validation of the childbirth experience questionnaire (CEQ-SK) in Slovakia, Eur J Midwifery (2023) 14:7:6.
- 101. Annborn A., Finnbogadóttir H.R., *Obstetric violence a qualitative interview study*, Midwifery 105 (2022) 103212, p. 5; Report of the Special Rapporteur on violence against women, its causes and consequences, Dubravka Šimonović, *A human rights-based approach to mistreatment and violence against women in reproductive health services with a focus on childbirth and obstetric violence*, A/74/137 (2019), paras. 39-41; PACE, *Obstetrical and gynaecological violence*, Resolution 2306 (2019) and Explanatory Memorandum by Ms Maryvonne Blondin, rapporteur, Doc. 14965, 16 Sept. 2019.
- 102. See footnote 97 and accompanying sources. See also European Court of Human Rights, *Y.P. v. Russia*, Application No. 43399/13, judgment of 20 September 2022, available at https://hudoc.echr.coe.int/?i=001-219209 (the case concerns sterilisation without free and informed consent during a delivery via Caesarean section).
- 103. Council of Europe Commissioner for Human Rights, *Addressing the invisibility of women and girls with disabilities*, Human Rights Comment published on 21 April 2022, available at https://www.coe.int/es/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities.
- 104. Council of Europe Commissioner for Human Rights, Issue paper on *Protecting the right to health through inclusive and resilient health care for all*, published by the (2021), pp. 5, 20-21, available at https:/rm.coe.int/protecting-the-right-to-health-through-inclusive-and-resilient-health-/1680a177ad.

- 105. Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual and reproductive health and rights in Europe*, published by the (2017), pp. 55-56, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead.
- 106. See, e.g., CRPD, Concluding Observations: Norway, CRPD/C/NOR/CO/1 (2019), paras. 8, 10, 40(c); Concluding Observations: Greece, CRPD/C/GRC/CO/1 (2019), paras. 10, 37(b); Concluding Observations: Germany, CRPD/C/DEU/CO/2-3 (2023), paras. 12(c), 52; CEDAW, Concluding Observations: Georgia, CEDAW/C/GEO/CO/6 (2023), para. 42; CERD, Concluding Observations: Portugal, CERD/C/PRT/CO/18-19 (2023), para. 14; CESCR, Concluding Observations: Czechia, E/C.12/CZE/CO/3 (2022), paras. 13(a), 15, 39, 41. Women's and girls' sexual and reproductive health rights in crisis, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), para. 77; Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, Tlaleng Mofokeng, Sexual and reproductive health rights: challenges and opportunities during the COVID-19 pandemic, A/76/172 (2021), para. 21.
- 107. Guaranteeing sexual and reproductive health and rights for all women, in particular women with disabilities, Joint statement by the Committee on the Rights of Persons with Disabilities and the Committee on the Elimination of All Forms of Discrimination against Women, 29 August 2018; CRC, Concluding Observations: Croatia, CRC/C/HRV/CO/5-6 (2022), para. 35(b); Concluding Observations: Poland, CRC/C/POL/CO/5-6 (2021), para. 36(b). See also Council of Europe Commissioner for Human Rights, Addressing the invisibility of women and girls with disabilities, Human Rights Comment published on 21 April 2022, available at https://www.coe.int/en/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities.
- 108. See, e.g., CERD, Concluding Observations: Slovakia, CERD/C/SVK/CO/11-12 (2018), para. 24; Concluding Observations: Slovakia, CERD/C/SVK/CO/13 (2022), para. 29; Concluding Observations: Croatia, CERD/C/HRV/CO/9-14 (2023), para. 22(e); Concluding Observations: Italy, CERD/C/ITA/CO/21 (2023), para. 27; CEDAW, Concluding Observations: North Macedonia, CEDAW/C/MKD/CO/6 (2018), para. 38(a); Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), para. 43(b); CESCR, Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), para. 45; HRC, Concluding Observations: Romania, CCPR/C/ROU/CO/5 (2017), para. 26.
- 109. See, e.g., CEDAW, Concluding Observations: North Macedonia, CEDAW/C/MKD/CO/6 (2018), para. 38(e); Concluding Observations: Serbia, CEDAW/C/SRB/CO/4 (2019), para. 38(d); CRC, Concluding Observations: Bosnia and Herzegovina, CRC/C/BIH/CO/5-6 (2019), para. 35(d); Concluding Observations: Poland, CRC/C/POL/CO/5-6 (2021), para. 36(b).
- 110. CRC, Concluding Observations: Greece, CRC/C/GRC/CO/4-6 (2022), para. 18(a); Concluding Observations: Cyprus, CRC/C/CYP/CO/5-6 (2022), para. 18(f); CESCR, Concluding Observations: Spain, E/C.12/ESP/CO/6 (2018), para. 42; Concluding Observations: Denmark, E/C.12/DNK/CO/6 (2019), para. 63; Concluding Observations: Slovakia, E/C.12/SVK/CO/3 (2019), paras. 46-47.
- 111. CEDAW, Concluding Observations: United Kingdom of Great Britain and Northern Ireland, CEDAW/C/GBR/CO/8 (2019), para. 56(b); HRC, Concluding Observations: Belgium, CCPR/C/BEL/CO/6 (2019), para. 30.
- 112. Women's and girls' sexual and reproductive health rights in crisis, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), para. 20; Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, Tlaleng Mofokeng, Sexual and reproductive health rights: challenges and opportunities during the COVID-19 pandemic,

- A/76/172 (2021), para. 23; Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, Tlaleng Mofokeng, *Violence and its impact on the right to health*, A/HRC/50/28 (2022); CEDAW, General recommendation No. 37 (2018) on the gender-related dimensions of disaster risk reduction in the context of climate change, CEDAW/C/GC/37 (2018), paras. 65-68.
- 113. Council of Europe Commissioner for Human Rights, Issue paper on *Women's sexual and reproductive health and rights in Europe* (2017), p. 42, available at https:/rm.coe.int/women-s-sexual-and-reproductive-health-and-rights-in-europe-issue-paper/168076dead.
- 114. See, e.g., Council of Europe Commissioner for Human Rights, *Addressing the rights of women with disabilities in Europe: observations and perspectives*, Speech (2022), available at https://www.coe.int/es/web/commissioner/-/addressing-the-rights-of-women-with-disabilities-in-europe-observations-and-perspectives; "Addressing the invisibility of women and girls with disabilities," Human Rights Comment, (2022), available at https://www.coe.int/es/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities; "Nothing to cure: putting an end to so-called "conversion therapies" for LGBTI people," Human Rights Comment (2023), available at https://www.coe.int/es/web/commissioner/-/nothing-to-cure-putting-an-end-to-so-called-conversion-therapies-for-lgbti-people; Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to the Czech Republic from 20 to 24 February 2023, CommHR(2023)26, available at https://rm.coe.int/report-on-czech-republic-by-dunja-mijatovic-commissioner-for-human-rig/1680ac88cd.
- 115. CEDAW, S.N. and E.R. v. North Macedonia, CEDAW/C/75/D/107/2016 (2020); L.A. et al v. North Macedonia, CEDAW/C/75/D/110/2016 (2020); S.B. and M.B. v. North Macedonia, CEDAW/C/77/D/143/2019 (2020); CERD, Concluding Observations: Slovakia, CERD/C/SVK/CO/11-12 (2018), paras. 23-24; Gheorghe C., Mocanu C., Challenging intersectionality: Roma women's voices and experiences Experiences of discrimination of Roma women in housing, education, health and employment Comparative research on multiple discrimination in Finland, Italy and Romania (2021), E-Romnja Association, Romni Onlus, Helsinki Deaconess Institute Foundation, Center for Not-for-profit Law, available at https://e-romnja.ro/en/download/2393/?tmstv=1694441952.
- 116. See, e.g., European Roma Rights Centre, Cause of Action: Reproductive Rights of Romani Women in Hungary (2020), available at http://www.errc.org/uploads/upload\_en/ file/5228 file1 reproductive-rights-of-romani-women-in-hungary.pdf; Cause Action: Reproductive Rights of Romani Women in Bulgaria (2020), available at http:// www.errc.org/uploads/upload\_en/file/5229\_file1\_reproductive-rights-of-romaniwomen-in-bulgaria.pdf; Cause of Action: Reproductive Rights of Romani Women in Ukraine (2019), available at http://www.errc.org/uploads/upload\_en/file/5126\_file1\_ cause-of-action-reproductive-rights-of-romani-women-in-ukraine.pdf; Joint submission to the Committee on the Elimination of Discrimination against Women concerning shortcomings in the implementation of the Convention in Slovakia, Fenestra, Women's Circles and Center for Civil and Human Rights (Poradňa) (April 2023), available at https://tbinternet.ohchr.org/\_layouts/15/treatybodyexternal/ Download.aspx?symbolno=INT%2FCEDAW%2FCSS%2FSVK%2F52464&Lang=en; CEDAW, Concluding Observations: Slovakia, CEDAW/C/SVK/CO/7 (2023), paras. 42(b)-43(b).
- 117. Council of Europe Commissioner for Human Rights, *Slovak Parliament has a historic opportunity to deliver justice to victims of forced sterilisation* (12.6.2023), available at <a href="https://www.coe.int/en/web/commissioner/-/slovak-parliament-has-a-historic-opportunity-to-deliver-justice-to-victims-of-forced-sterilisation">https://www.coe.int/en/web/commissioner/-/slovak-parliament-has-a-historic-opportunity-to-deliver-justice-to-victims-of-forced-sterilisation</a>; Center for Civil

- and Human Rights (Poradňa), *Our NGO calls on the Slovak Parliament to address the shortcomings of the currently negotiated draft law on compensation for illegally sterilized women* (20.6.2023), available at https://poradna-prava.sk/en/news/our-ngo-calls-on-the-slovak-parliament-to-address-the-shortcomings-of-the-currently-negotiated-draft-law-on-compensation-for-illegally-sterilized-women/; Call on the Minister of Justice to ensure justice for women who have been forcibly sterilised in the past (19.12.2023), available at https://poradna-prava.sk/wp-content/uploads/2023/12/Vyzva-MVO-Poradna-na-zabezpecenie-spravodlivosti-pre-zeny-ktore-boli-v-minulosti-sterilizovane-v-rozpore-s-pravom-.pdf; Committee against Torture, Concluding Observations: Slovakia, CAT/C/SVK/CO/4 (2023), paras. 23-24.
- 118. Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to the Czech Republic from 20 to 24 February 2023, CommHR(2023)26, sec. 2.2., available at https://rm.coe.int/report-on-czech-republic-by-dunja-mijatovic-commissioner-for-human-rig/1680ac88cd; Czech Health Ministry refusing evidence other than original medical records when awarding compensation for forced sterilizations, more than 1,500 victims have applied, Romea.cz (15.11.2023), available at https://romea.cz/en/czech-republic/czech-health-ministry-refusing-evidence-other-than-original-medical-records-when-awarding-compensation-for-forced-sterilizations-more-than-1500-victims-have-applied; Odškodňování za protiprávní sterilizace romských žen je třeba výrazně zjednodušit a zrychlit, shodla se romská vládní zmocněnkyně, osobnosti i neziskovky, Romea.cz (10.12.2023), available at https://romea.cz/cz/domaci/odskodnovani-za-protipravni-sterilizace-romskych-zen-je-treba-vyrazne-zjednodusit-a-zrychlit-shodla-se-romska-vladni-zmocnenkyne-osobnosti-i-neziskovky.
- 119. See, e.g., CEDAW, S.N. and E.R. v. North Macedonia, CEDAW/C/75/D/107/2016 (2020); L.A. et al v. North Macedonia, CEDAW/C/75/D/110/2016 (2020); S.B. and M.B. v. North Macedonia, CEDAW/C/77/D/143/2019 (2020); Gheorghe C., Mocanu C., Challenging intersectionality: Roma women's voices and experiences Experiences of discrimination of Roma women in housing, education, health and employment Comparative research on multiple discrimination in Finland, Italy and Romania (2021), E-Romnja Association, Romni Onlus, Helsinki Deaconess Institute Foundation, Center for Not-for-profit Law, page 94-115, available at https://e-romnja.ro/en/download/2393/?tmstv=1694441952.
- 120. Council of Europe Commissioner for Human Rights, *Addressing the invisibility of women and girls with disabilities*, Human Rights Comment published on 21 April 2022, available at https://www.coe.int/es/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities.
- 21. Council of Europe Commissioner for Human Rights, *Addressing the invisibility of women and girls with disabilities*, Human Rights Comment published on 21 April 2022, available at <a href="https://www.coe.int/en/web/commissioner/-/addressing-the-rights-of-women-with-disabilities-in-europe-observations-and-perspectives">https://www.coe.int/en/web/commissioner/-/addressing-the-rights-of-women-with-disabilities-in-europe-observations-and-perspectives</a>; European Court of Human Rights, *G.M. and Others v. the Republic of Moldova*, Application No. 44394/15, judgment of 22 November (2022); CRPD, Concluding Observations: Georgia, CRPD/C/GEO/CO/1 (2023), paras. 35-36; Concluding Observations: Hungary, CRPD/C/HUN/CO/2-3 (2022), paras. 35-36; Turkey, CRPD/C/TUR/CO/1 (2019), paras. 36-37; European Disability Forum, *A silenced truth stories of forced sterilisation in the EU* (2023), available at <a href="https://www.edf-feph.org/publications/a-silenced-truth-stories-of-forced-sterilisation-in-the-eu/">https://www.edf-feph.org/publications/a-silenced-truth-stories-of-forced-sterilisation-in-the-eu/</a>.
- 122. Council of Europe Commissioner for Human Rights, *Addressing the invisibility of women and girls with disabilities*, Human Rights Comment published on 21 April 2022, available at https://www.coe.int/es/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities.

- 123. Council of Europe Commissioner for Human Rights, *Addressing the invisibility of women and girls with disabilities*, Human Rights Comment published on 21 April 2022, available at https://www.coe.int/es/web/commissioner/-/addressing-the-invisibility-of-women-and-girls-with-disabilities; European Disability Forum, *Sexual and reproductive health and rights of women and girls with disabilities* (2019), available at https://www.edf-feph.org/content/uploads/2021/02/edf\_position\_paper\_on\_srhr\_english\_0.pdf; ILGA-Europe, European Disability Forum, *Intersections: The LGBTI II Survey persons with disabilities analysis* (2023), available at https://www.ilga-europe.org/report/intersections-disabilities-diving-into-the-fra-lgbti-ii-survey-data/; CRPD, Concluding Observations: Luxembourg, CRPD/C/LUX/CO/1 (2017), para. 44(b), 45(b).
- 124. See, e.g., IPPF, International Medical Advisory Panel IMAP Statement on sexual and reproductive health and rights of the ageing population (2018); AGE Platform Europe, Too Old for Sexual Life?, Press Release (8.3.2019), available at https://www.age-platform.eu/too-old-for-sexual-life/; World Elder Abuse Awareness Day: Ways Forward (14.6.2022), available at https://www.age-platform.eu/world-elder-abuse-awareness-day-ways-forward/; ILGA-Europe, AGE Platform Europe, Intersections: The LGBTI II Survey older people analysis (2023), available at https://www.ilga-europe.org/report/intersections-older-people-diving-into-the-fra-lgbti-ii-survey-data/; Report of the Independent Expert on the enjoyment of all human rights by older persons, Claudia Mahler, Violence against and abuse and neglect of older persons, A/HRC/54/26 (2023), paras. 42-43.
- 125. See, e.g., Center for Reproductive Rights, *Perilous Pregnancies: Barriers in Access to Affordable Maternal Health Care for Undocumented Migrant Women in the European Union* (2020), available at https://reproductiverights.org/wp-content/uploads/2021/03/Updated-GLP-EUROPE-PerilousPregnancies-2020-Web.pdf; Council of Europe Commissioner for Human Rights, Dunja Mijatović, Report following her visit to Spain from 21 to 25 November 2022, CommHR(2023)9, para. 56, available at https://rm.coe.int/report-on-spain-following-the-visit-from-21-to-25-november-2022-by-dun/1680aaeb17; PICUM, *Migration status: A key structural social determinant of health inequalities for undocumented migrants* (2023), available at https://picum.org/wp-content/uploads/2023/12/Migration-status\_A-key-structural-social-determinant-of-health-inequalities-for-undocumented-migrants\_EN.pdf.
- 126. See, e.g., Center for Reproductive Rights, *Perilous Pregnancies: Barriers in Access to Affordable Maternal Health Care for Undocumented Migrant Women in the European Union* (2020), available at <a href="https://reproductiverights.org/wp-content/uploads/2021/03/Updated-GLP-EUROPE-PerilousPregnancies-2020-Web.pdf">https://reproductiverights.org/wp-content/uploads/2021/03/Updated-GLP-EUROPE-PerilousPregnancies-2020-Web.pdf</a>; PICUM, *Data Protection and the "Firewall": Advancing the Right to Health for People in an Irregular Situation* (2020), available at <a href="https://picum.org/wp-content/uploads/2020/07/PICUM-Briefing\_Data-protection-and-the-firewall\_health.pdf">health.pdf</a>; CESCR, Concluding Observations: Germany, E/C.12/DEU/CO/6 (2018), paras. 26-27; Report of the Special Rapporteur on the human rights of migrants, *The impact of migration on migrant women and girls: a gender perspective*, A/HRC/41/38 (2019), para. 60.
- 127. European Court of Human Rights, *R.R. and Others v. Hungary*, Application No. 36037/17, judgment of 2 March 2021, paras. 52, 58, 65; *V.M. and Others v. Poland*, Application No. 40002/22, communicated on 10 January 2023.
- 128. See, e.g., *Taking the pulse: A Prescription for Equality and Enhanced Healthcare Access for LGBTI people in Europe*, Council of Europe SOGI Newsroom (24.11.2023), available at https://www.coe.int/en/web/sogi/-/taking-the-pulse-a-prescription-for-equality-and-enhanced-healthcare-access-for-lgbti-people-in-europe; ILGA-Europe, *Health4LGBTI: Reducing health inequalities experienced by LGBTI people*, (9.3.2018), available at https://www.ilga-europe.org/campaign/health4lgbti-reducing-health-inequalities-experienced-by-lgbti-people/.

- 129. Council of Europe Commissioner for Human Rights, *Nothing to cure: putting an end to so-called "conversion therapies" for LGBTI people*, Human Rights Comment published on 16 February 2023), available at https://www.coe.int/en/web/commissioner/-/nothing-to-cure-putting-an-end-to-so-called-conversion-therapies-for-lgbti-people.
- 130. See, e.g., European Committee of Social Rights, Transgender Europe and ILGA-Europe v. the Czech Republic, Complaint No. 117/2015, decision of 15 May 2018; European Court of Human Rights, *X. and Y. v. Romania*, Applications No. 2145/16 et 20607/16, judgment of 19 January 2021; ILGA-Europe, TGEU, *Intersections: The LGBTI II Survey Trans and Non-Binary Analysis* (2023), available at https://www.ilga-europe.org/report/intersections-trans-non-binary-diving-into-the-fra-lgbti-ii-survey-data/.
- 131. See, e.g., ILGA-Europe, Oll Europe, *Intersections: The LGBTI II Survey Intersex Analysis* (2023), available at https://www.ilga-europe.org/report/intersections-intersex-diving-into-the-fra-lgbti-ii-survey-data/; HRC, Concluding Observations: Finland, CCPR/C/FIN/CO/7 (2021), paras. 20-21; Concluding Observations: Germany, CCPR/C/DEU/CO/7 (2021), paras. 20-21; Concluding Observations: Luxembourg, CCPR/C/LUX/CO/4 (2022), paras. 11-12; Concluding Observations: Ireland, CCPR/C/IRL/CO/5 (2023), paras. 19-20; CAT, Concluding Observations: Netherlands, CAT/C/NLD/CO/7 (2018), paras. 52-53; CRPD, Concluding Observations: Switzerland, CRPD/C/CHE/CO/1 (2022), paras. 35(c), 36(c). See also Council of Europe Commissioner for Human Rights, *Issue paper on Human rights and intersex people*, (2015), available at http://rm.coe.int/human-rights-and-intersex-people-issue-paper-published-by-the-council-/16806da5d4.
- 132. See, e.g., Fertility Europe, European Parliamentary Forum for Sexual and Reproductive Rights, *European Atlas of Fertility Treatment Policies* (2021), available at https://fertilityeurope.eu/atlas/.
- 133. Council of Europe Commissioner for Human Rights, *Protecting the human rights of sex workers*, Human Rights Comment published on 15 February 2024, available at https://www.coe.int/en/web/commissioner/-/protecting-the-human-rights-of-sex-workers.
- 134. See, e.g., Report of the Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health, Tlaleng Mofokeng, Sexual and reproductive health rights: challenges and opportunities during the COVID-19 pandemic, A/76/172 (2021), paras. 10, 12; The Special Rapporteur on the right of everyone to the enjoyment of the highest attainable standard of physical and mental health's Written Submissions to the European Court of Human Rights in the case M.A and Others v. France (Applications No. 63664/19, 64450/19, 24387/20, 24391/20 et 24393/20), 30 September 2021; Eliminating discrimination against sex workers and securing their human rights, Working Group on discrimination against women and girls, A/HRC/WG.11/39/1 (2023).
- 135. Council of Europe Commissioner for Human Rights, *Human rights defenders in the Council of Europe area in times of crises*, CommHR(2023)2, page 12, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813.
- 136. See for example, Council of Europe Commissioner for Human Rights, Human rights defenders in the Council of Europe area in times of crises, CommHR(2023)2, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813; No space for violence against women and girls in the digital world (2022), Human Rights Comment published on 15 March 2022, available at https://www.coe.int/en/web/commissioner/-/no-space-for-violence-against-women-and-girls-in-the-digital-world; PACE, Access to abortion in Europe: stopping anti-choice harassment, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022; Center for Reproductive Rights et al., Care in Crisis: Failures to Guarantee the Sexual and Reproductive Health and Rights of Refugees

- from Ukraine in Hungary, Poland, Romania and Slovakia (2023), pp. 8, 61-74, available at https://reproductiverights.org/ukraine-report-care-in-crisis/; Amnesty International, An unstoppable movement: A global call to recognize and protect those who defend the right to abortion (2023), available at https://www.amnesty.org/en/documents/pol40/7420/2023/en/.
- 137. See, e.g., PACE, Access to abortion in Europe: stopping anti-choice harassment, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022; Amnesty International, Legal but not local: Barriers to accessing abortion services in Northern Ireland (2023), available at https://www.amnesty.org.uk/barriers-abortion-access-northern-ireland-0.
- 138. See, e.g., Council of Europe Commissioner for Human Rights, *Human rights defenders in the Council of Europe area in times of crises*, CommHR(2023)2, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813; *Women's and girls' sexual and reproductive health rights in crisis*, Report of the Working Group on discrimination against women and girls, A/HRC/47/38 (2021), paras. 41-42, 79.
- 139. PACE, Access to abortion in Europe: stopping anti-choice harassment, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022; Council of Europe Commissioner for Human Rights, Human rights defenders in the Council of Europe area in times of crises, CommHR(2023)2, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813; Human rights of LGBTI people in Europe: current threats to equal rights, challenges faced by defenders, and the way forward, CommDH(2021)32, available at https://rm.coe.int/human-rights-of-lgbti-people-in-europe-current-threats-to-equal-rights/1680a4be0e; Human Rights Defenders in the Council of Europe Area: Current Challenges and Possible Solutions, CommDH(2019)10, paras. 23-25, available at https://rm.coe.int/hr-defenders-in-the-coe-area-current-challenges-and-possible-solutions/168093aabf; Center for Reproductive Rights et al., Care in Crisis: Failures to Guarantee the Sexual and Reproductive Health and Rights of Refugees from Ukraine in Hungary, Poland, Romania and Slovakia (2023), pp. 61-67, available at https://reproductiverights.org/ukraine-report-care-in-crisis/.
- 140. Council of Europe Commissioner for Human Rights, *Protect and enable the work of human rights defenders contributing to ensuring access to safe abortion care in Europe*, Statement of 28 September 2023, available at https://www.coe.int/en/web/commissioner/-/protect-and-enable-the-work-of-human-rights-defenders-contributing-to-ensuring-access-to-safe-abortion-care-in-europe.
- 141. *Polish abortion: Activist guilty of providing pills to end pregnancy*, BBC News (14.3.2023), available at https://www.bbc.com/news/world-europe-64950505. The conviction is being appealed.
- 142. Andorra: Acquittal of Human Rights Defender Vanessa Mendoza Cortes, Amnesty International, the Center for Reproductive Rights, Women's Link Worldwide and Front Line Defenders, Press Release (17.1.2024), available at https://reproductiverights.org/andorra-acquittal-of-human-rights-defender-vanessa-mendoza-cortes/; Council of Europe Commissioner for Human Rights, Andorra: guarantee the right to freedom of expression for women's rights defender Vanessa Mendoza Cortés and ensure an enabling environment for women's rights defenders, Statement (29.11.2023), available at https://www.coe.int/es/web/commissioner/-/andorra-guarantee-the-right-to-freedom-of-expression-for-women-s-rights-defender-vanessa-mendoza-cortés-and-ensure-an-enabling-environment-for-women-s-rights-defenders.
- 143. Council of Europe Commissioner for Human Rights, *Human rights defenders in the Council of Europe area in times of crises*, CommHR(2023)2, para. 17, available at https://

- rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813. See also PACE, *Access to abortion in Europe: stopping anti-choice harassment*, Resolution 2439 (2022) and Explanatory memorandum by Ms Margreet De Boer, rapporteur, Doc. 15459, 21 Feb. 2022, para. 54.
- 144. Council of Europe Commissioner for Human Rights, *Human rights defenders in the Council of Europe area in times of crises*, CommHR(2023)2, para. 37, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813.
- 145. Center for Reproductive Rights et al., Care in Crisis: Failures to Guarantee the Sexual and Reproductive Health and Rights of Refugees from Ukraine in Hungary, Poland, Romania, and Slovakia (2023), pp. 64, 69, available at https://reproductiverights.org/ukraine-report-care-in-crisis/.
- 146. Council of Europe Commissioner for Human Rights, *Human rights defenders in the Council of Europe area in times of crises*, CommHR(2023)2, para. 39, available at https://rm.coe.int/report-on-the-round-table-human-rights-defenders-in-the-council-of-eur/1680aaa813.

The Commissioner for Human Rights is an independent and impartial non-judicial institution established in 1999 by the Council of Europe to promote awareness of and respect for human rights in the member states.

The activities of this institution focus on three major, closely related areas:

- country visits and dialogue with national authorities and civil society,
- thematic studies and advice on systematic human rights work, and
- awareness-raising activities.

The current Commissioner, Dunja Mijatović, took up her funtions in April 2018. She succeeded Nils Muižnieks (2012-2018), Thomas Hammarberg (2006-2012) and Álvaro Gil-Robles (1999-2006).



www.commissioner.coe.int

**ENG** 

#### www.coe.int

The Council of Europe is the continent's leading human rights organisation. It comprises 46 member states, including all members of the European Union. All Council of Europe member states have signed up to the European Convention on Human Rights, a treaty designed to protect human rights, democracy and the rule of law. The European Court of Human Rights oversees the implementation of the Convention in the member states.



